Corporate social responsibility, market reaction and accounting conservatism

Accounting conservatism

Received 23 January 2020 Revised 8 June 2020 Accepted 7 July 2020

Xixi Shen

School of Finance, Nankai University, Tianjin, China

Kung-Cheng Ho

Pearl River Delta Collaborative Innovation Center of Scientific Finance and Industry, Institute of Regional Finance, Guangdong University of Finance and Economics, Guangdong, China

Lu Yang

College of Economics, Shenzhen University, Guangdong, China, and

Leonard Fong-Sheng Wang

Wenlan School of Business, Zhongnan University of Economics and Law, Wuhan, China

Abstract

Purpose — Non-financial information disclosure may reflect the quality of corporate financial reports or disclosure policy choices. The authors examine the relationship between corporate social responsibility (CSR) and accounting conservatism and also investigate channels through which such effects are transmitted. The purpose of this paper is to explore how CSR, as non-financial information that has received widespread attention, affects choices regarding corporate financial policy.

Design/methodology/approach — Using ordinary least squares regression, the authors analyze China CSR Score data for 2010–2018. They control certain influencing variables related to the nature and characteristics of enterprises and discover that CSR can effectively increase accounting conservatism. Then, they extract the components of market reactions to CSR and study the market reaction path of CSR as it affects financial policy. They also conduct a robustness test to ensure that the results are not accidental in a complex environment.

Findings – The results reveal the influence of non-financial information on firms' financial policy. In addition, the results confirm the attraction of liquidity and investor attention as the major market reaction channels by which CSR significantly promotes accounting conservatism. Additionally, other critical paths of influence deserve further exploration. The results remain robust for alternate measures of accounting conservatism, different components of CSR, other proxies on CSR, endogenous testing and alternate estimation methods.

Originality/value – The study represents the first analysis of the influence of CSR information disclosure on accounting conservatism in emerging markets, and it undertakes a preliminary exploration to clarify the mechanism of CSRs' role in accounting conservatism. The results also provide a policy reference for external supervision and internal governance of enterprises. Thus, the results can help company managers maintain a favorable corporate image and establish a high-level investor protection mechanism.

Keywords CSR, Accounting conservatism, Investor protection, Investor attention, Stock liquidity **Paper type** Research paper



JEL classification – G32, G38

Funding: The authors were funded by NSSFC number (19ZDA061, 19AJY027), NSFC number (71903199)

Kybernetes © Emerald Publishing Limited 0368-492X DOI 10.1108/K-01-2020-0043

1. Introduction

Accounting conservatism and corporate social responsibility (CSR) have long been topics of concern in their respective fields. With high concern but little research, the study of the interaction between these two topics is the focus of our discussion. In the early stages of development, the rough economic development model has focused excessively on the speed of economic growth and has ignored the quality of economic growth. Numerous countries have realized that balancing economic growth and social responsibility is critical. Previous studies have revealed that high standards of CSR disclosure can help corporate managers establish a positive image, achieve optimal performance, reduce the risk of stock price crash and increase their stock price (Kim et al., 2014). CSR affects not just the development of the company itself but also sustainable global development and the interests of the public. For example, the shocking Chinese milk powder contamination incident in 2008 not only ruined the future of the enterprises involved but also seriously endangered public safety and damaged public trust in domestic milk powder. Another example is the "European Plastics Strategy" for plastic products in the EU introduced in 2018. This requires the reduction of plastic raw materials and restrictions on plastic products, which effectively controls plastic pollution and encourages green consumption. This strategy has become a crucial method for enterprises to respond to environmental problems and fulfill their social responsibilities while also contributing to environmental protection. Accounting conservatism is another long-term practice that eases the flow of company-specific information from insiders to outsiders and leads to a high-quality information environment (Hu et al., 2014). Because listed companies are more inclined to report positive events and not report negative developments and because investors are typically more sensitive to losses than to gains, accounting conservatism helps investors to understand negative operational developments of listed companies. It also plays a crucial role in corporate information disclosure and investor protection. Our research revolves around these two critical topics, as we explore the influence of corporate CSR on accounting conservatism and the channels of this influence.

CSR is a hot topic and numerous thorough studies have been published on the subject. This research has affected the corporate CSR disclosure policy in numerous ways. Dolores Sánchez-Fernández et al. (2014) studied the relationship between institutional theory and CSR in the hotel industry and discovered that values and norms created by regulatory pressures, organizational activities and the organizational pursuit of legality have a positive effect on the adoption of CSR practices. The corporate image that results from highly developed CSR, increased participation in CSR activities and the perception of legality can promote CSR disclosure (Michaels and Grüning, 2018). However, managers deciding whether to undertake CSR activities must also consider the corresponding cost of capital. Huang et al. (2016) discussed how two complementary manufacturers should decide between CSR and costs when confronted with customer requirements for product safety assurances and the incorporation of CSR into their corporate philosophies. They also discussed the effects of increasing the degree of CSR in the production process. On the other hand, CSR has also been gradually applied to corporate reputation and business strategy, and thus, it has received even more research attention. The relevant literature has mainly focused on the relationship between CSR and two factors:

- The relationship between enterprises and stakeholders and the relationship between corporate investment and financing.
- (2) Financial information disclosure and other factors.

CSR plays a critical role in improving stakeholder relationships and investment efficiency while reducing investment costs. Huang and Watson (2015) argued that although CSR

mainly appears to focus on non-shareholders, it can also enhance a company's reputation, brand and trust; attract customers and employees and ultimately improve profitability and company value. Thus, high-quality CSR disclosure can help achieve maximum shareholder value. However, numerous studies have argued that although mandatory CSR disclosure can change a company's pollutant emission behavior and can produce positive externalities, it can also reduce corporate profits and damage shareholders' interests (Chen et al., 2018). Additionally, it can also greatly reduce information asymmetry between companies and creditors and mitigate conflicts of interest between managers and different stakeholders (Becchetti et al., 2012; Nguyen et al., 2017; Yang et al., 2018). Gong and Ho (2018a) revealed that market competition provides modest incentives for managers to engage in CSR behavior that benefits all stakeholders. Digital communication tools have changed how companies interact with external stakeholders and the public. Illia et al. (2017) studied how companies use new media to engage with stakeholders on matters related to CSR. They discovered that companies typically engage in dialogue with stakeholders, but the dialogue is dominated by the company; only a few companies discuss the concerns of stakeholders and the wider public. Investors can use the internet to obtain information regarding merchants' and brands' reputations to strengthen their trust (Goldfarb and Tucker, 2019). Rivat (2016) argued that the introduction of digital technology into the distribution of CSR goals can provide improved results for achieving sustainable social growth. CSR also plays a major role in creating a single digital market, which creates an open, fair and seamless network environment and removes market barriers (Georgeta et al., 2016). In the context of digital transformation, corporate digital responsibility integrates moral considerations at corporate, personal and social levels to provide a new direction to guide the relationship between labor and technology and to shape the ethical use of new technologies (Orbik and Zozuľaková, 2019). Furthermore, digital technology can significantly increase its users' understanding of financial information and improve their financial decision-making ability (French et al., 2019). This also promotes increased requirements for corporate financial information disclosure. In terms of capital allocation, high levels of CSR result in low information asymmetry, which can improve investment efficiency and reduce investment sensitivity to Tobin's Q in a variety of ways (Bhandari and Javakhadze, 2017; Benlemlih and Bitar, 2018). Therefore, high-quality CSR disclosures can also reduce a company's cost of equity capital and lead to lower loan costs, a lower risk premium on corporate bonds and increased ability to raise debt during a potential crisis (El Ghoul et al., 2011; Dhaliwal et al., 2014; Lins et al., 2017; Cheung et al., 2018; Gong et al., 2018). Ee et al. (2018) discovered that there is no such effect in the short term. Dhaliwal et al. (2011) revealed that increasing equity capital and disclosing CSR activities that promote each other. CSR improves the relationships between managers and shareholders, creditors and other stakeholders, which reduces agency issues and conflicts of interest. Furthermore, CSR can increase investment efficiency and attract funds at a relatively low cost due to the reduction in information asymmetry that it provides.

Another research aspect of CSR is the relationship between CSR information disclosure and financial information disclosure. Because of the ubiquity of CSR in modern enterprises, an increasing number of companies report CSR information on a regular basis. Numerous studies have revealed that companies with strong CSR reputations are more likely to disclose high-quality financial information than those with low CSR reputations (Labelle et al., 2010; Kim et al., 2012). Additionally, Huang and Watson (2015) confirmed the existence of a natural connection between CSR and accounting, in which accounting is responsible for measurement, disclosure and assurance of all information, including CSR-related information. CSR begins with self-regulation and in some cases lacks a formal regulatory

structure, so accounting is crucial in CSR reporting. Gong and Ho (2018b) revealed that monitoring CSR reports reduces information asymmetry and effectively limits managerial short-termism. CSR information disclosure is affected by accounting, and it also counteracts financial information, affects the quality of financial information and improves the overall information disclosure environment. CSR research also covers many aspects, firm value (El Ghoul et al., 2011; Albuquerque et al., 2019; Gong et al., 2019), business performance and profitability (Becchetti et al., 2012; Chen et al., 2016; Lins et al., 2017; Byun and Oh, 2018), Chief Executive Officer (CEO) characteristics (Jian and Lee, 2015; Petrenko et al., 2016; McCarthy et al., 2017; Tang et al., 2018) and family control and institutional ownership (El Ghoul et al., 2016; Dyck et al., 2019). These studies have revealed that the relationship between businesses and stakeholders, financing, investment and financial information are closely related to CSR and that CSR is also affected by many factors. In short, CSR, as a critical non-financial factor, is closely related to all aspects of a company and is inextricably linked to financial information.

Accounting policies profoundly affect the quality of accounting information disclosure. Accounting conservatism refers to the asymmetry of positive and negative disclosures, with delayed recognition of benefits and accelerated recognition of losses (Basu, 1997). The most intuitive effect of accounting conservatism is that net assets and accumulated net income are under-reported relative to neutral or aggressive accounting methods. The research on accounting conservatism has mainly investigated two questions:

- (1) whether accounting conservatism should be implemented; and
- (2) the role of accounting conservatism.

Two perspectives have been presented on whether the principle of accounting conservatism should be enforced. First, both the Financial Accounting Standards Board and International Accounting Standards Board have promoted the transition from conservative accounting principles to fair value accounting because of the belief that conservatism can bias accounting information, and thus, affect information neutrality. Some researchers have agreed with the view that conservative, non-neutral financial statements negatively affect the quality of financial reporting, which, in turn, leads to inefficient decision-making (Gigler et al., 2009; Guay and Verrecchia, 2006). For example, underestimating net income and net asset value can affect a firms' operational performance ratios. If investors cannot effectively distinguish between underestimated ratios and actual ratios, decision bias might easily result.

However, some scholars have argued that accounting conservatism as a longstanding accounting principle is uniquely superior to accounting neutrality. Watts (2003) elaborated on the asymmetric verification requirement of accounting conservatism from several aspects. He argued that accounting conservatism effectively limits opportunistic managerial behavior and incentives to introduce bias and noise into valuation estimates and that it also reduces net assets and opportunistic payments to managers and other parties, which increases the value of the company and the welfare of all stakeholders. Conservatism also helps to alleviate the possibility of excessive distribution of wealth by stakeholders. LaFond and Watts (2008) argued that accounting conservatism can reduce the ability and incentives of managers to manipulate financial results, thus alleviating information asymmetry to some extent. The academic research has proven that conservatism includes debt value in accounting information and also reduces the information asymmetry between borrowers and lenders, thus benefiting both lenders and borrowers (Ahmed *et al.*, 2002; Ball *et al.*, 2008a). Ball *et al.* (2008b) analyzed the syndicated loan market and revealed that conservatism increases transparency and helps to improve the efficiency of lending

contracts. Juan *et al.* (2014) agreed that increased conservatism leads to a subsequent reduction in information asymmetry and discovered that conservatism is not only useful to debt holders but also shareholders. Equity investors use accounting information to assess the value of stocks and make investment decisions. Additionally, accounting information can help the parties to a contract to assess the effectiveness and efficiency of contract performance obligations such as debt levels and executive compensation. To a large extent, the two different perspectives on accounting conservatism depend on the use of the accounting information because not all information is valuable to both investors and parties to a contract.

As for the role of accounting conservatism, it reduces information asymmetry and agency costs and protects investors. Young (2005) argued that moral hazard-related agency costs, which derive from incentives that lead to the implementation of suboptimal decisions, can be reduced by accounting conservatism. Jenkins et al. (2009) noted that during economic contractions, corporate profitability declines and management and auditors tend to report financial information more conservatively. This both reduces the information asymmetry between management and external stakeholders, which helps to attract more external financing and reduces litigation risk. Highly accurate and reliable disclosure can protect investors, and the disclosure of negative news is more informative and more representative of corporate value than non-disclosure of such negative news (Langberg and Sivaramakrishnan, 2008). According to Hu et al. (2014), accounting conservatism helps specific information flow to outsiders, resulting in a high-quality information environment. Moreover, accounting conservatism can also replace the legal system in ensuring information quality, thus effectively protecting the interests of investors. In other words, accounting conservatism can also protect investors to some extent. However, strong investor protection can also reduce corporate earnings management and conceal a company's actual performance from the external view (Leuz et al., 2003). Francis and Wang (2008) discovered that as investor protections become stronger, earnings quality and accounting conservatism of companies audited by Big-four auditors' increase. In summary, accounting conservatism effectively improves the information disclosure environment and investor protection. We summarize the literature on CSR and accounting conservatism in Table 1.

Our research is divided into two steps. The first step focuses on the relationship between CSR and accounting conservatism. CSR information is beyond the scope of financial disclosures. However, it requires the support of accounting and also supplements financial reports, which implies that it is closely related to accounting information. Thus, a discussion of how CSR relates to the quality of financial reporting may have interesting findings. Limited research has been conducted on the direct relationship between CSR and accounting conservatism. Chih *et al.* (2008) revealed that an increased commitment to CSR leads to reductions in the degree of earnings smoothing and the avoidance of disclosing losses and decreases in earnings, but it also leads to an increase in earnings aggressiveness.

The second step of our research explores the path by which CSR promotes accounting conservatism. We follow the market reaction channel of Hou and Moskowitz (2005) propose two channels between investor attention and liquidity. Investor attention may be a channel by which accounting conservatism is promoted. Brennan and Tamarowski (2000) suggested that companies attract analyst coverage through highly transparent disclosures and financial statements that reduce the cost of information acquisition, and it has a positive effect on stock liquidity. After the disclosure of mandatory CSR reports, the number of analysts following a company increases and market information asymmetry declines, especially for companies with low initial analyst coverage (Hung *et al.*, 2013). Lin (2016)

Table 1. Literature form

Research topics	Authors	The research found
CSR		
Stakeholder relations	Becchetti et al. (2012)	Mitigate conflicts
	Huang and Watson (2015)	Maximum shareholder value
	Nguyen <i>et al.</i> (2017)	Create shareholder value
	Yang et al. (2018)	Reduce information asymmetry
	Gong and Ho (2018a)	±
Investment		Reduce investment sensitivity to Tobin's
mvestment	Benlemlih and Bitar (2018)	+
Capital cost	El Ghoul <i>et al.</i> (2011)	T
Capital Cost	Dhaliwal <i>et al.</i> (2011)	_
		_
	Dhaliwal <i>et al.</i> (2014)	
	Ye and Zhang (2011)	U-shape
	Lins et al. (2017)	Increase the ability to raise the debt
	Cheung <i>et al.</i> (2018)	_
	Gong <i>et al.</i> (2018)	_
Financial information	Kim <i>et al.</i> (2012)	+
disclosure	Huang and Watson (2015)	Have a natural connection
	Gong and Ho (2018b)	Limits managerial short-termism
Characteristics of CEO	Jian and Lee (2015)	+
	Petrenko et al. (2016)	+
	McCarthy et al. (2017)	<u>-</u>
	Tang <i>et al.</i> (2018)	_
Business performance/	Aupperle <i>et al.</i> (1985)	_
profitability/firm value	McWilliams and Siegel (2000)	_
promability/mm value	Lins <i>et al.</i> (2017)	+
	Byun and Oh (2018)	+
	Chen <i>et al.</i> (2018)	-
	Albuquerque <i>et al.</i> (2019)	+
	Gong <i>et al.</i> (2019)	+
Conservatism		
Efficiency of contracts	Guay and Verrecchia (2006)	_
Efficiency of contracts	Ball <i>et al.</i> (2008b)	+
Information environment	Ahmed <i>et al.</i> (2002)	+
information environment	Guay and Verrecchia (2006)	T
	Ball <i>et al.</i> (2008a)	+
		+
	Ball et al. (2008b)	+
	Gigler <i>et al.</i> (2009)	_ +
	Jenkins <i>et al.</i> (2009)	+
	Juan <i>et al.</i> (2014)	+
	Hu et al. (2014)	+
Beneficiary	Watts (2003)	Stakeholders
	Juan <i>et al.</i> (2014)	Debt holders and shareholders
	Hu <i>et al.</i> (2014)	Investors
Managerial behavior	Watts (2003)	Limit opportunistic behavior
	LaFond and Watts (2008)	Reduce manipulation behavior
CCD 1 ti		
CSR and conservatism	1777 (0000)	T
Mandated CSR	Kurniawan and Wibowo (2009)	Insignificant
	Karsalari <i>et al.</i> (2017)	_
	Cheng and Kung (2016)	+
Voluntary CSR	Pyo and Lee (2013)	+
Mandated and voluntary CS	R Salewski and Zulch (2014)	_
Note: This table provides a	summary of the literature on CSR a	nd accounting concervation

proved that the level of institutional investor ownership is closely related to accounting conservatism. Short-term institutional investors who hold diversified portfolios have a high portfolio turnover rate, focus on short-term investment returns and trade on current earnings news. Therefore, as such transient institutional investor ownership increases, firms exhibit lower accounting conservatism. By contrast, increases in dedicated, long-term institutional ownership lead to increased accounting conservatism. Therefore, investor attention may be the first channel by which CSR affects accounting conservatism. Information disclosure may directly affect market liquidity, and thus, also lead to accounting conservatism. Disclosure policies affect market liquidity, and a transparent disclosure policy can reduce information asymmetry and effectively reduce transaction costs, thereby increasing the liquidity of the company's stock (Welker, 1995; Heflin et al., 2005). Chung et al. (2010) discovered that companies can improve their stock liquidity by adopting corporate governance standards that mitigate information asymmetry. Thus, we assume that the second medium of CSRs' influence on accounting conservatism is stock liquidity. In summary, we draw on Hou and Moskowitz (2005) to define two influence paths by which CSR affects accounting conservatism. The first path is CSRs' ability to attract investor attention and the second path is its ability to increase stock liquidity.

Our research has three main contributions. First, whereas prior research on CSR and accounting conservatism has been limited, we study the effect of CSR on accounting conservatism, thus enriching the relevant literature. Our exploration of two potential market reaction channels represents the first such survey in the literature, and thus, expands upon prior research. We use the ordinary least squares method to extract the investor attention component, the liquidity component and the remaining component of CSR when we examine CSR market reaction channels that affect accounting conservatism. This method clearly distinguishes the different channels of influence, and it is both simple and practical. Our study clarifies the influence mechanism of CSR and accounting conservatism and it reveals that CSR has a positive effect on accounting conservatism. Additionally, the results remain robust when we apply alternate accounting conservatism measures and various CSR measures. Investor attention is the major path by which CSR affects accounting conservatism. Specifically, high CSR scores affect firms' accounting conservatism by attracting investors. Liquidity is another critical channel of influence. Our findings also reveal that other critical channels of influence exist, in addition to investor attention and stock liquidity. Second, many countries have realized the importance of balancing the relationship between CSR and economic development. However, compared with developed countries, much room remains for the improvement of CSR in developing countries. China is among the worlds' most dynamic economies and has developed rapidly, but its level of CSR development is not high, the supervision of CSR disclosure started late and its CSR disclosure exhibits certain representativeness. Research on CSR in China, despite its low base, has developed rapidly (Moon and Shen, 2010). We study the influence of CSR on accounting conservatism in the Chinese market, which has critical implications for understanding emerging markets. Moreover, the high-quality CSR rating indicators that we use to cover many aspects of social responsibility; these indicators are widely used in research and practice and can accurately evaluate CSR performance. They are, thus highquality scoring indicators, which increases the accuracy of our study. Third, CSR information is non-financial reporting information and a supplement to the shortcomings of financial disclosure. It also reduces information asymmetry, and its quality can also reflect the quality of financial information. We analyze the effect of non-financial reporting information on the characteristics and quality of financial reporting information, which facilitates users' assessments of the conservativeness of financial statements, increases their

ability to obtain accurate financial numbers and helps them to make optimal decisions. At the same time, accounting conservatism is also related to investor protection. Our research provides a basis for investors to identify the characteristics and quality of enterprises' financial information disclosure. It also provides a new way of evaluating the level of investor protection.

The remainder of this paper is organized as follows. Section 2 discusses the related literature and develops our hypothesis. Section 3 describes our data and variable construction. Section 4 describes the empirical methodology and presents our main results. Section 5 presents our discussions and implications. Section 6 concludes the paper.

2. Literature review and hypothesis development

In this section, we establish assumptions to test the effect of firms' CSR information disclosure quality on accounting conservatism and the channel of this effect.

Corporate management and governance and information disclosure ratings are interdependent. Excellent profitability and operating conditions can increase a company's level of voluntary disclosure (Healy and Palepu, 2001; Francis *et al.*, 2008). High levels of disclosure can significantly reduce a company's cost of capital (Leuz and Verrecchia, 2000; Hail, 2002; Lambert *et al.*, 2007; Shroff *et al.*, 2013), affect a company's financing decisions (Pan *et al.*, 2015) and increase its stock value, returns and investor expectations of future performance (Jiao, 2011). Companies with relatively high board independence tend to provide comprehensive payroll disclosure, thereby reducing agency conflicts (Sheu *et al.*, 2010). Chung *et al.* (2015) revealed that voluntary disclosure of comprehensive information can alleviate the negative relationship between excess executive compensation and company value. Hermalin and Weisbach (2012) noted that disclosure and other governance reforms can be a double-edged sword. Although increased corporate information can improve the ability of shareholders and board directors to oversee management, mandatory increases in disclosure partly explain the recent increase in CEO compensation and CEO turnover.

Bushman and Smith (2001) defined the governance role of financial accounting information as the use of externally reported financial accounting data in a control mechanism that promotes effective corporate governance. They argued that financial accounting information can influence how a firm invests. Additionally, CSR can influence the choice of accounting policies (Bozzolan *et al.*, 2015). Financial information and nonfinancial information can complement each other, and the effect of poor accounting quality can be offset by other sources of non-accounting information that can serve as a substitute mechanism for a company to maintain information transparency (Callen *et al.*, 2013). Dhaliwal *et al.* (2011) revealed that financial and CSR disclosures can replace each other in reducing the cost of equity capital. Once companies report their CSR performance, they have a stronger motivation than previously to improve the quality of their financial statements because of ethical issues (Kim *et al.*, 2012). Bereskin *et al.* (2018) framed CSR as a critical component of a company's culture and used the similarity of CSR characteristics as a proxy variable for cultural similarity. In addition, corporate financial information disclosure is affected by a company's culture.

Modern accounting standards emphasize non-financial indicators as a means of compensating for weaknesses in financial indicators. Non-financial information supplements financial information and can also affect financial information disclosure. Therefore, we propose H1:

H1. CSR helps to enhance accounting conservatism.

Accounting

Further analysis of the influence channels is necessary to understand the effect of CSR information disclosure on accounting conservatism. We mainly study the intermediary effects of investor attention and stock liquidity. CSR information and financial statement information complement and influence each other, which results in both increased information disclosure quality and timeliness. Regina (2008) revealed that conservative reporting reduces borrower information asymmetry and increases the efficiency of secondary debt securities transactions. Dyck et al. (2019) revealed that investor requirements can promote the environmental and social performance of global companies, and financial motivations and cultural origin are critical for improving environmental and social performance for investors.

Most scholars have argued that increased levels of disclosure attract additional analyst coverage, whereas reduced disclosure has the opposite effect. Analysts are critical market participants who gather and interpret stock information so that investors can engage in more informed trading decisions than would otherwise be the case. Large, global companies voluntarily provide higher levels of forward-looking and historical non-financial disclosures than other companies, which effectively and improves the accuracy of financial analysts' earnings forecasts (Vanstraelen et al., 2003). Lee et al. (2018) examined analyst recommendations from an informational perspective and discovered that a company's voluntary disclosure of CSR-related information or the release of CSR-related information by third-party rating agencies can cause a decline in the value of stock analysis and recommendations. As such, CSR information reduces the influence of financial analysts on price discovery by reducing the degree of information opacity, which helps investors make informed decisions using their judgment. Tsao et al. (2016) studied the relationship between voluntary monthly earnings disclosure and analyst coverage, and they discovered that companies that disclose monthly income attract more analyst coverage than companies that do not disclose monthly income. Diamond and Verrecchia (1991) argued that public information disclosure reduces information asymmetry and attracts large investors, thereby increasing the liquidity of securities and reducing the company's capital cost, Core (2001) summarized the relevant literature and argued that analysts and institutions generate information, which reduces information asymmetry; but for informed investors, they are more inclined to reduce disclosure. Dhaliwal et al. (2011) revealed that companies with outstanding social responsibility performance attract specialized institutional investors and analyst coverage than companies with weak social responsibility performance. Analyst coverage and increased institutional ownership are also associated with increased liquidity in corporate stocks (Roulstone, 2003; Boone and White, 2015). Healy et al. (1999) discovered that increased disclosure evaluations lead to increased stock returns, institutional ownership, analyst coverage and stock liquidity. However, Beyer et al. (2010) noted that not all companies' stock liquidity improves when analyst coverage increases, and the causal relationship between analyst coverage and stock liquidity is also unclear. Benlemlih and Bitar (2018) also studied the benefits of CSR and revealed that companies with higher CSR participation have lower information asymmetry than other companies. Furthermore, CSR can effectively shape corporate investment behavior and improve investment efficiency. Albuquerque et al. (2019) revealed that consumers are more crucial determinants of a company's CSR policy than investors, but not all CSR activities are aimed at promoting customer loyalty. Employee loyalty also has a critical influence on CSR. CSR information disclosure can attract investor attention, improving the company's information environment and increasing its liquidity.

In addition to increasing liquidity through investor attention, information disclosure can increase liquidity directly. An *et al.* (2012) examined panel data on real estate investment

trusts and discovered that information asymmetry plays a critical role in corporate liquidity management. Liquidity interacts with corporate governance. Specifically, liquidity improves corporate governance, information disclosure and corporate value. Corporate governance can increase corporate transparency, reduce adverse selection and attract increased transactions (i.e. liquidity) in the company's stock. This demonstrates that when corporate governance quality improves, the stock liquidity is significantly improved (Prommin *et al.*, 2014). Ng (2011) revealed that improved information quality is associated with reduced liquidity risk, and when a substantial effect on market liquidity occurs, the negative correlation between information disclosure quality and liquidity risk is stronger than when such an effect does not occur.

In summary, we argue that CSR, investor attention, accounting conservatism and stock liquidity are inextricably linked. Thus, we follow the market reaction channel of Hou and Moskowitz (2005) propose *H2*:

H2. CSR further contributes to increased accounting conservatism the market reaction channel.

3. Data and variable construction

The data on CSR in our study comes from a third-party agency, Rankins CSR ratings (RKS). The RKS database covers the 2009–2018 period. However, due to the incomplete collection of CSR information in 2009, our sample comprises the performance of all listed common stocks on the Shanghai Stock Exchange (SHSE) and the Shenzhen Stock Exchange (SZSE) in 2010–2018. Other data on the Chinese stock market comes from the China Stock Market financial database – statement notes of China stock market and accounting research. Considering the availability of data, we exclude the following from our sample: companies that lack CSR reports and companies that lack enough disclosed information to measure accounting conservatism. Following these exclusions, 4,472 observations are included in our sample.

3.1 Corporate social responsibility

According to a 2008 survey by the Chinese Academy of Social Sciences, CSR is a concern among most investors. The Chinese government realizes the importance of balancing economic growth and social responsibility and has issued a series of relevant guidelines. In 2007, the State-owned Assets Supervision and Administration Commission of the State Council issued the "guidelines for central CSR information disclosure." In January 2009, the China Banking Industry Association issued the "guidelines for CSR of Chinese Banking Financial Institutions." In September 2018, the China Securities Regulatory Commission issued the revised "guidelines for corporate governance of listed companies." The revision strengthened the leading role of listed companies in social responsibility and established a basic framework for disclosure of CSR information.

In addition, China's two major exchanges have launched CSR guides. On September 25, 2006, the SZSE issued the "guidelines for social responsibility of listed companies," which mandated the inclusion of CSR reports attached to annual disclosure reports of SZSE 100 index components and encouraged other enterprises to voluntarily disclose CSR reports. On May 14, 2008, the SHSE strengthened the required social responsibility information disclosure and issued the "guidelines for environmental information disclosure of listed companies," which required three types of companies (i.e. components of the SHSE corporate governance sector index, financial companies and listed overseas joint-stock

companies) to provide CSR reports. In December 2008, the SHSE and the SZSE jointly issued the "notice on doing a good job in the 2008 annual report of listed companies." In 2009, the SHSE launched the CSR Index.

CSR requires people to pay attention to corporate profits, but it also requires systematic thinking to make decisions and to take actions on a very broad basis (Knez-Riedl *et al.*, 2006). However, interpretations of CSR in the literature have varied and no single consistent definition has been commonly accepted. The CSR performance data that we examine is from the RKS database, which measures CSR by four dimensions, namely, the macrocosm, content, technique and industry dimensions. Integrity refers to the effectiveness of CSR strategies, the effectiveness of management governance and the evaluation of various stakeholders. Content refers to the evaluation of social responsibility in terms of community, environment, product strategy and sustainable social development. The technique dimension considers the relative balance of content and related innovation capabilities. The industry dimension is an evaluation of a company's compliance with relevant industry norms, RKS publishes CSR rating scores ranging from 1 to 100 during the 2009–2018 period and is an effective measure of CSR among Chinese companies. Thus, the data is applicable for empirical research (Gong and Ho, 2018a, Gong and Ho, 2018b). With the increasing importance of CSR and the recognition of the quality of the CSR data in the RKS database, the number of CSR reports disclosed has gradually increased over the years. In 2009, 371 listed companies issued CSR reports and that number grew to 851 by 2018.

Table 2 lists and describes the main variables used in our study (Table 3).

3.2 Accounting conservatism

Accounting conservatism is a critical feature of accounting information quality, and typically imposes higher requirements for positive news disclosure than for negative news disclosure. This can result in undervalued net assets (Basu, 1997). We apply two measures of accounting conservatism. The first measure is the level of accounting conservatism at the annual level as described by Khan and Watts (2009). We construct this measure of accounting conservatism based on the cross-sectional regression model of Basu (1997):

$$X_i = \beta_0 + \beta_1 D_i + \beta_2 R_i + \beta_3 D_i R_i + \varepsilon_i \tag{1}$$

among which X_i is the earnings per share divided by the beginning-of-period stock price; R_i is the stock return of the company in the respective year; D_i is a dummy variable that is 1 when R_i is less than zero and is otherwise 0; and ε_i is a random error term. The coefficient β_2 indicates the timeliness of positive news disclosure, the coefficient β_3 indicates the degree of improvement in the timeliness of negative news disclosure relative to positive news disclosure (i.e. the level of conservatism) and $\beta_2 + \beta_3$ indicates the timeliness of the overall negative news disclosure. The annual-enterprise level factor or the timeliness of positive news confirmation β_2 and accounting conservatism β_3 , can be expressed as a linear function, as follows:

$$GScore = \beta_2 = \mu_0 + \mu_1 SIZE_i + \mu_2 MB_i + \mu_3 LEV_i$$
 (2)

$$CScore = \beta_3 = \omega_0 + \omega_1 SIZE_i + \omega_2 MB_i + \omega_3 LEV_i$$
(3)

where $SIZE_i$ is the natural logarithm of total assets, MB_i is the ratio of the market value of equity to book value and LEV_i is the firms' leverage ratio. Equations (2) and (3) are not

Variables	Variable definitions
Accounting con	nservatism measures
CON_KIM	See equation (1) as follow Kim and Zhang (2016)
CON KHAN	See equation (2) as follow Khan and Watts (2009)
_	
CSR measures	
CSR	The CSR score ranging from 1 (the lowest) to 100 (the highest)
CSR_{ATT}	The CSR component of investor attention channel
CSR_{LIQ}	The CSR component of investor liquidity channel
CSR_{error}	$CSR_{error} = CSR - CSR_{ATT} - CSR_{LIQ}$
CSRS	The standardized CSR score, defined as the difference between CSR and the average value of CSR, scale
	by the standard deviation of CSR for firm i at time t
$CSRS_{ATT}$	The CSRS component of investor attention channel
$CSRS_{LIQ}$	The CSRS component of investor liquidity channel
$CSRS_{error}$	$CSRS_{error} = CSRS - CSRS_{ATT} - CSRS_{LIQ}$
Cortoerror	Cond _{error} - Conto Conto _{ATT} Conto _{LlQ}
Investor attent	
IO	Institutional ownership (%) = stock ownership of foreign institutions, domestic funds and securities
	companies
ANA	A unique number of analysts providing earnings forecasts
EMP	A number of employees
SH	A number of shareholders
ADV	The natural log of $(1 + advertising expense)$. Adv is set to zero when advertising expense is missing
Liquidity meas	wes
PRC	Average daily share price
TRD	A total number of trading days
ILR	The natural log of Amihud (2002)'s illiquidity ratio defined as average daily absolute return divided by
ILI	dollar trading volume
	ariables measures
CI	Capital intensity defined as the ratio of property, plant and equipment to total assets
OM	Operating margin defined as sales minus cost of goods sold scaled by sales
OC	The operating cycle defined as average receivables divided by sales plus average inventory divided by
	cost of goods sold
BIG4	The big-4 auditor is a dummy variable that has a value of one of the firm has a Big-4
Control variab	les measures
AZ	Altman's Z-score, defined as (3.3*operating income + sales + 1.4*retained earnings + 1.2*(current assets -
112	current liability)/total assets
CFV	The standard deviation of cash flows over the past three years
DUAL	CEO duality: a dummy variable, with 0 for a company having separate CEO and chairman and 1
CDOHMAN	otherwise
GROWTH	Percentage change in sales in the fiscal year
LEV	The total long-term debt divided by total assets
<i>LISTAGE</i>	Firm age, measured by the natural logarithm of $(1 + \text{the firm's established period})$
LOSS	The relative frequency of losses in the previous three years (the number of loss years divided by three).
	loss year is one in which net income before extraordinary items is negative
MB	The ratio of the market value of equity to book value of equity
ROA	Income before extraordinary items divided by lag total assets for the fiscal year preceding the investment
	date/period
SIGMA	The standard deviation of firm-specific weekly returns over the fiscal year period
SIZE	Market capitalization (billions)
SOE	A dummy variable that equals 1 if the ultimate controlling shareholder of a listed firm in the state, and
OOL	otherwise
TOP (%)	Percentage of total outstanding shares owned by the largest shareholder
. 7	<u> </u>
Note: This t	able defines each dependent and the independent variable used in the empirical analysis

Table 2. Variable definitions

Variable	Mean	SD	First (%)	First quartile	Median	Third quartile	99th (%)	Accounting conservatism
Accounting cons								
CON_KIM	0.12	0.14	-0.09	0.02	0.08	0.21	0.45	
CON_KHAN	-0.84	2.96	-10.15	-0.13	0.01	0.15	0.76	
CSR measures	00.00	11.10	10.00	20.40	05.04	41.00	20.54	
CSR	36.62	11.18	13.33	29.40	35.04	41.92	68.74	
CSRS	-0.07	0.82	-1.29	-0.64	-0.27	0.26	2.56	
Investor attentio	n measur							
IO	1.53	0.81	0.00	0.97	1.66	2.12	3.05	
ANA	2.05	1.17	0.00	1.10	2.30	3.04	3.89	
EMP	1.88	1.41	0.14	0.90	1.54	2.42	8.11	
SH	3.62	1.32	0.00	2.94	3.71	4.53	6.21	
ADV	0.25	1.09	0.00	0.00	0.00	0.00	6.25	
Liquidity measu	res							
PRC	2.69	0.61	1.33	2.25	2.67	3.09	4.16	
TRD	5.44	0.10	5.02	5.46	5.48	5.49	5.50	
ILR	-6.97	1.02	-8.73	-7.76	-7.07	-6.35	-4.09	
Instrumental va	riahles							
CI	0.23	0.17	0.00	0.09	0.20	0.33	0.69	
OM	0.29	0.17	0.01	0.17	0.27	0.38	0.76	
OC	0.82	1.13	0.03	0.25	0.45	0.87	6.70	
BIG4	0.08	0.28	0.00	0.00	0.00	0.00	1.00	
Control variables	c							
AZ	3.25	5.71	-0.06	1.25	2.53	4.08	13.77	
CFV	0.05	0.25	0.00	0.01	0.03	0.05	0.22	
DUAL	0.20	0.40	0.00	0.00	0.00	0.00	1.00	
GROWTH	0.12	0.29	-0.55	-0.02	0.11	0.26	0.96	
LEV	0.22	0.40	0.00	0.00	0.05	0.33	1.54	
LISTAGE	2.28	0.63	0.77	1.78	2.44	2.83	3.15	
LOSS	0.12	0.32	0.00	0.00	0.00	0.00	1.00	
MB	2.15	2.19	0.13	0.83	1.49	2.70	10.32	
ROA	0.06	0.06	-0.10	0.02	0.04	0.08	0.24	
SIGMA	0.05	0.02	0.02	0.04	0.05	0.06	0.11	
SIZE	0.09	0.20	0.00	0.01	0.03	0.09	0.88	
SOE	0.46	0.50	0.00	0.00	0.00	1.00	1.00	
TOP (%)	35.54	16.41	7.93	21.84	34.05	47.33	76.07	

Notes: The summary statistics of the main variables. Corporate social responsibility score includes *CSR* is the CSR score ranging from 1 (the lowest) to 100 (the highest) and *CSRS* is the standardized CSR score, defined as the difference between CSR and the average value of CSR, scaled by the standard deviation of CSR for firm *i* at time *t*. *Accounting conservatism measure* is accounting conservatism level include *CON_KIM* and *CON_KHAN*. Investor attention variables including institutional ownership (*IO*), analyst coverage (*ANA*), number of employees (*EMP*) and number of shareholders (*SH*), advertising expense (*ADV*) and stock liquidity variables including share price (*PRC*), number of trading days (*TRD*) and Amihud (2002) illiquidity ratio (*ILR*). Instrumental variables such as capital intensity (*CI*), operating margin (*OM*), operating cycle (*OC*) and Big-four auditor dummy (*BIG4*). Control variables include Altman's Z-score (*AZ*), volatility of cash flows (*CFV*), CEO duality (*DUAL*), sales revenue growth rate (*GROWTH*), firm's leverage (*LEV*), firm age (*LISTAGE*), relative loss frequency (*LOSS*), market value of equity to book value of equity (*MB*), the return of asset (ROA), volatility (*SIGMA*), firms' size (*SIZE*), state-owned enterprise (*SOE*) and largest shareholder ratios (*TOP* %). Table 2 provides details of all the main variables

Table 3. Summary statistics

estimation equations. We combine equations (2) and (3) into equation (1) to obtain regression equation (4) to estimate accounting conservatism, as follows:

$$\begin{split} X_{i} &= \beta_{0} + \beta_{1}D_{i} + R_{i}(\mu_{0} + \mu_{1}SIZE_{i} + \mu_{2}MB_{i} + \mu_{3}LEV_{i}) \\ &+ D_{i}R_{i}(\omega_{0} + \omega_{1}SIZE_{i} + \omega_{2}MB_{i} + \omega_{3}LEV_{i}) \\ &+ (\vartheta_{1}SIZE_{i} + \vartheta_{2}MB_{i} + \vartheta_{3}LEV_{i} + \vartheta_{4}D_{i}SIZE_{i} + \vartheta_{5}D_{i}MB_{i} + \vartheta_{6}D_{i}LEV_{i}) + \varepsilon_{i} \end{split}$$

$$(4)$$

We estimate equation (4) on a year-by-year basis and obtain the coefficients ω_0 , ω_1 , ω_2 and ω_3 , which we input into equation (3) to obtain the annual-enterprise accounting conservatism index.

Our second indicator of accounting conservatism is based on Kim and Zhang (2016), who used a five-year rolling panel to estimate the CScore indicator. This estimation method, which is based on the improved Basu (1997) model, uses panel data to estimate accounting conservatism. The coefficient can be changed for each examined company and time period, and the indicators that are obtained are highly reliable. Therefore, the empirical research in this paper mainly focuses on this index. Companies with higher CScores are considered to be more conservative in their accounting than companies with lower CScores.

3.3 Market reaction variables

We select five variables to properly measure investor attention between 2010 and 2018, namely, institutional ownership (*IO*), analyst attention (*ANA*), number of employees (*EMP*), number of shareholders (*SH*) and advertising expenses (*ADV*). Institutional ownership comprises stock ownership by foreign institutions, domestic funds and securities companies. We summarize the number of analysts (or teams) who followed the company in each year as a measure of analyst interest. The number of employees refers to the number of employees in service in any given year. The number of shareholders refers to the number of shareholders in each year, reflecting that the degree of market competition and the company's equity capital costs are significantly affected by asymmetric information (Armstrong *et al.*, 2011). Advertising costs are the amount spent on advertising in each calendar year. The specific variables are defined in Table 2.

Referring to Hou and Moskowitz (2005), we define the natural logarithm (*PRC*) of the stock price, the natural logarithm of the trading day (*TRD*) and the illiquidity ratio of Amihud (2002), for which we adopt the natural logarithm value (*ILR*), as our liquidity variable. Specifically, stock price refers to the average daily stock price during a calendar year and the number of trading days is the sum of the stocks' actual trading days during each year. Amihud's (2002) illiquidity ratio is defined as the daily average absolute rate of return divided by the trading volume (in millions of Chinese yuan) during a given year.

3.4 Instrumental variables

Healy and Palepu (2001) argued that companies with strong operating performance tend to disclose higher-quality information than companies with weak operations. Companies with superior fundamentals may also have a higher quality of disclosure and higher accounting conservatism than those with weak fundamentals. Thus, the disclosure rating may be an endogenous variable that causes a deviation from the final result. Following the steps suggested by Larcker and Rusticus (2010), we perform endogenous testing, the results of which indicate that endogeneity is a problem that is worth considering. To alleviate the possible problem of endogeneity, we consider economic theory and select the six

instrumental variables proposed by Larcker and Rusticus (2010), the natural logarithm of the number of common shareholders, sales growth in a given year, capital intensity, operating margin, length of the operating cycle and a dummy variable that indicates whether the firm is audited by a Big-four accounting firm. Next, we evaluate the instrumental variables to check for problems of overidentification and weak instrumental variables. The null hypothesis of the overidentification test is that all instrumental variables are exogenous. In the weak instrumental variable test, a Kleibergen–Paap Wald F statistic that is greater than 10 indicates that no weak instrumental variable problem exists. Finally, four eligible instrumental variables are selected, namely, capital intensity (CI), operating margin (OM), length of the operating cycle (OC) and whether the firm is audited by a Big-four accounting firm (BIG4). The specific variable definitions are presented in Table 2.

3.5 Control variables

To ensure that the results are accurate, we also control other variables that may affect accounting conservatism. Referring to the relevant literature, these variables comprise Altman's Z-score (AZ) (Nikolaev, 2010; Lara *et al.*, 2016), cash flow volatility (CFV) (Lara *et al.*, 2016), sales revenue growth rate (GROWTH) (Ahmed and Duellman, 2013), firm leverage (LEV) (Watts, 2003; Khan and Watts, 2009; Juan *et al.*, 2014; Kim and Zhang, 2016), relative loss frequency (LOSS) (Lara *et al.*, 2016), return on assets (ROA) (Nikolaev, 2010; Kim and Zhang, 2016) and volatility (SIGMA) (Kim and Zhang, 2016). We also control company characteristics such as the shareholding ratio of the largest shareholder, board structure (Lara *et al.*, 2007), market-to-book ratio (Watts, 2003; Khan and Watts, 2009; Juan *et al.*, 2014; Kim and Zhang, 2016), firm size (Watts, 2003; Khan and Watts, 2009; Juan *et al.*, 2014; Kim and Zhang, 2016) and firm age. The corporate governance structure of China's state-owned enterprises (SOEs) is very different from Western SOEs and non-SOEs, so we also control for whether a firm is an SOE (Gong and Ho, 2018a). Again, the specific variable definitions are presented in Table 2.

3.6 Average firm characteristics of corporate social responsibility separated portfolios Table 4 presents the average firm characteristics of CSR in separate portfolios. In our classification, the CSR scores were divided into Groups 1 to 5, from low to high. A clear difference exists between Groups 1 and 5. We discover that companies with higher CSR scores tend to have higher accounting conservatism than companies with lower CSR scores. The difference between CON KIM of the highest score group and the lowest score group is 0.08 and significantly. The investor attention indicators exhibit statistically significant differences between groups, indicating that the value of the high-CSR score group is significantly larger than that of the low-CSR score group. In other words, the values of the investor attention variables for companies with high CSR scores are larger, and such companies tend to have higher levels of institutional ownership, more analyst coverage, more employees and shareholders and higher advertising costs than companies with low CSR scores. Among the liquidity measures, only the Amihud (2002) illiquidity ratio exhibits significant differences between the groups. Compared to companies with low CSR scores, companies with high CSR scores tend to be more value-oriented (MB), more regulated in terms their corporate governance (DUAL) and larger (SIZE); their largest shareholder also tends to have a higher shareholding ratio (TOP). These firms include a greater number SOEs, and they tend to have longer firm ages (LISTAGE), higher leverage (LEV) and higher profitability (ROA) than companies with low CSR scores.

	1	2	3	4	5	5–1
CSR rank						
CSR	25.65	31.03	34.48	39.49	52.53	26.88 (38.55)***
CSRS	-0.93	-0.56	-0.27	0.15	1.24	2.17 (53.24)***
Accounting conse	rvatism meası					
CON_KIM	0.10	0.09	0.10	0.13	0.18	0.08 (5.67)***
CON_KHAN	-1.02	-0.77	-0.87	-0.83	-0.69	0.33 (1.20)
Investor attention	measures					***
IO	1.42	1.45	1.58	1.62	1.60	0.18 (2.64)
ANA	1.79	1.95	2.08	2.12	2.34	0.55 (5.71)***
<i>EMP</i>	1.35	1.46	1.67	2.28	2.63	1.28 (11.70)
SH	3.43	3.51	3.64	3.55	3.99	0.56 (5.07)***
ADV	0.11	0.03	0.36	0.19	0.59	0.48 (4.29)***
Liquidity measure	es					
PRC	2.67	2.71	2.67	2.74	2.64	-0.03(-0.60)
TRD	5.44	5.45	5.44	5.44	5.45	0.01 (1.10)
ILR	-7.22	-7.17	-6.92	-6.91	-6.61	0.61 (6.98)***
Instrumental vari	iables					
CI	0.21	0.24	0.23	0.23	0.25	0.04 (2.71)***
OM	0.30	0.29	0.28	0.30	0.28	-0.02(-1.23)
OC	0.98	0.82	0.82	0.69	0.78	$-0.20(-1.82)^*$
BIG4	0.05	0.02	0.06	0.09	0.22	0.17 (6.31)***
Control variables						
AZ	3.23	2.98	3.95	3.25	2.83	-0.39(-0.65)
CFV	0.07	0.07	0.04	0.05	0.04	-0.02(-0.77)
DUAL	0.23	0.21	0.22	0.20	0.16	$-0.06(-1.87)^*$
GROWTH	0.14	0.10	0.12	0.12	0.13	-0.01(-0.34)
LEV	0.22	0.21	0.21	0.18	0.29	0.06 (1.91)*
<i>LISTAGE</i>	2.21	2.20	2.26	2.29	2.42	0.21 (3.80)***
LOSS	0.16	0.13	0.11	0.08	0.12	-0.04(-1.30)
MB	2.47	2.26	2.27	2.11	1.66	-0.81 (-4.29)***
ROA	0.05	0.05	0.06	0.06	0.06	0.01 (2.48)***
SIGMA	0.05	0.05	0.05	0.05	0.05	-0.01 (-4.84)***
SIZE	0.08	0.08	0.08	0.11	0.11	0.04 (2.61)***
SOE	0.37	0.39	0.46	0.54	0.59	0.22 (5.25)***
TOP (%)	33.99	35.02	35.73	36.52	36.55	2.56 (1.78)*

Table 4.Characteristics of CSR ranking-separated portfolios

In addition, our results reveal that companies with high CSR scores attract more investor attention than those with low scores. For example, the difference between the high-scoring and low-scoring companies is 0.55, and the difference between EMP is 1.28. The other variables of investor attention (i.e. IO, SH and ADV) are significantly larger in the high-scoring group than companies in the low-scoring group, which indicates that companies with high scores attract more investor attention.

Companies with high CSR scores also have higher stock liquidity than those with low scores. ILR differs by 0.61 between the highest-scoring group and the lowest-scoring group, and the *p* values are all less than 0.01.

3.7 Pearson correlation coefficients among variables

Table 5 exhibits the Pearson correlation coefficient matrix and variance inflation factors (VIF) between the variables used in our study. As expected, CSR (CSRS) exhibits a significant positive correlation with accounting conservatism CON_KIM (CON_KHAN); that is a relatively high CSR disclosure score is related to accounting conservatism. We also observe that both SIZE and SOE are positively correlated with CSR score (CSRS), but MB is negatively correlated with CSRS.

The higher a firms' CSR score, the more attention it attracts from investors. Among the investor attention indicators, IO, ANA, EMP, SH and ADV are all positively correlated with CSR, suggesting that companies with higher CSR scores attract more institutional shareholders, analyst coverage, employees and shareholders and such firm also tends to have higher advertising expenses than companies with lower CSR scores. Among the liquidity indicators, the correlation between TRD and CSR is negative, whereas the relationship between ILR and CSR is positive. All VIF in Table 5 are less than 6, indicating no serious linear correlation.

4. Methodology and empirical results

4.1 Corporate social responsibility and accounting conservatism

We run the regression illustrated in equation (5) to examine how a firms' CSR influences its accounting conservatism.

$$CON_{i,t} = \forall_0 + \forall_1 CSR_{i,t} + \sum_{i,t} \forall_{i,t} CONTROL_{i,t} + \varepsilon_{i,t}$$
(5)

where $CON_{i,t}$ is the enterprise accounting conservatism index; $CSR_{i,t}$ is the company's CSR score; and $CONTROL_{i,t}$ is a control variable that includes Altman's Z-score (AZ), cash flow volatility (CFV), CEO duality (DUAL), sales revenue growth rate (GROWTH), firm leverage (LEV), firm age (LISTAGE), relative loss frequency (LOSS), market-to-book ratio (MB), the total return on assets (ROA), volatility (SIGMA), firms' size (SIZE), SOE and largest shareholder ratio (TOP). The regression considers year and industry fixed effects. Table 2 provides details of all the variables.

Table 6 presents the results of this regression. We can see that CSRS significantly and positively explains accounting conservatism, illustrating that firms with higher CSR scores have higher accounting conservatism than those with lower CSR scores. After the control variables are controlled, the coefficients of CSRS in the regression of CON_KIM are 0.0014 (0.0178), both of which are significant at the 1% confidence level. The results above show that the higher a firms' CSR score, the higher its accounting conservatism, which implies that CSR disclosure helps to enhance accounting conservatism. As critical non-financial information, CSR disclosure can affect a firms' financial information disclosure policy. This result supports *H1* and it also supports the results of the Pearson correlation coefficient in Table 5. Moreover, it indicates a

(14)	1.00 0.23 0.03 0.03 0.06 0.08 0.06 0.07 0.11 0.55 0.07 0.11
(13)	1.00 -0.24 -0.40 0.02 -0.05 -0.08 0.07 0.17 -0.17 -0.13 -0.13
(12)	1.00 -0.16 0.03 0.21 0.26 -0.03 -0.01 0.14 0.15 0.05 -0.05 -0.05 -0.01 0.14 0.16 0.10
(11)	$\begin{array}{c} 1.00 \\ -0.04 \\ 0.02 \\ 0.03 \\ 0.03 \\ 0.03 \\ 0.03 \\ 0.04 \\ 0.00 \\ 0$
(10)	1.00 -0.09 -0.09 -0.03 -0.15 0.03 0.18 -0.18 -0.18 0.05 0.45 0.45 0.45 -0.14
(6)	1.00 -0.05 0.05 0.26 -0.19 0.25 0.25 0.25 0.03 0.25 0.03 0.05
(8)	1.00 0.19 0.08 0.06 0.47 0.00 0.02 0.02 0.03 0.03 0.03 0.03 0.03 0.03 0.03 0.03 0.03 0.03 0.04 0.04 0.04 0.04 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.07
(7)	$\begin{array}{c} 1.00 \\ -0.13 \\ 0.22 \\ 0.03 \\ 0.02 \\ 0.03 \\ 0.02 \\ 0.03 \\ 0$
(9)	1.00 0.01 0.37 0.33 0.01 0.01 0.02 0.02 0.03 0.04 0.02 0.03 0.03 0.03 0.03 0.03
(2)	1.00 0.20 0.20 0.10 0.00 0.00 0.00 0.00
(4)	1.00 0.16 0.16 0.34 0.16 0.23 0.04 0.04 0.07 0.00 0.00 0.00 0.00 0.00
(3)	1.00 0.00 0.00 0.00 0.31 0.01 0.02 0.04 0.04 0.04 0.01 0.01 0.01 0.01 0.01
(2)	1.00 0.31 0.07 0.07 0.03 0.03 0.03 0.04 0.04 0.04 0.03 0.04 0.05 0.05 0.05 0.06 0.06 0.07 0.07 0.07 0.07 0.07 0.03 0.03 0.03
(1)	1.00 0.22 0.23 0.03 0.03 0.03 0.03 0.03 0
VIF	5.54 5.50 5.50 5.50 5.50 5.50 5.50 5.50
Variables	(1) CON_KIM (2) CON_KHAN (3) CSR (4) CSRS (5) IO (6) ANA (7) EMP (8) SH (9) ADV (10) PRC (11) TRD (12) ILR (13) CC (14) OM (15) OC (16) BIG4 (17) AZ (18) CFV (19) DUAL (20) GROWTH (21) LEV (21) LEV (22) LISTAGE (23) LISTAGE (24) MB (25) ROA (26) SIGMA (27) SIZE (28) SOE (28) SOE (28) SOE (29) TOP

Notes: The table presents the Pearson's correlation matrix of all the variables used in the sample. Italic font denotes statistical significance at the 10% level (continued)

(53)	1.00	Accounting
(28)	$\frac{1.00}{0.14}$	conservatism
(27)	1.00 0.05 -0.10	
(26)	1.00 0.33 -0.16	
(25)	1.00 0.16 0.01 0.01	
(24)	1.00 0.36 0.33 0.32 -0.24	
(23)	$\begin{array}{c} 1.00 \\ -0.11 \\ -0.41 \\ 0.08 \\ 0.06 \\ 0.06 \\ 0.04 \end{array}$	
(22)	1.00 0.07 0.026 0.015 0.09 0.05	
(21)	$\begin{array}{c} 1.00 \\ 0.28 \\ 0.11 \\ -0.21 \\ -0.02 \\ 0.02 \\ 0.02 \\ 0.06 \\ 0.06 \end{array}$	
(20)	$\begin{array}{c} 1.00 \\ 0.04 \\ 0.05 \\ 0.07 \\ 0.07 \\ 0.02 \\ 0.07 \\ 0.007 \\ 0.007 \\ 0.002 $	
(19)	1.00 0.01 -0.08 -0.02 -0.05 0.06 -0.06	
(18)	1.00 -0.07 -0.03 0.04 0.04 0.01 0.01 0.01	
(17)	$\begin{array}{c} 1.00 \\ -0.02 \\ -0.05 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ 0.01 \\ 0.00 \\ $	
(16)	$\begin{array}{c} 1.00 \\ -0.01 \\ -0.05 \\ 0.00 \\ 0.04 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.09 \\ 0.01 \\ 0.04 \\ 0.01 \\ 0.04 \\ 0.01 \\ $	
(15)	$\begin{array}{c} 1.00 \\ -0.03 \\ -0.15 \\ -0.01 \\ 0.00 \\ 0.26 \\ 0.14 \\ 0.02 \\ 0.12 \\ -0.03 \\ -0.12 \\ -0.02 \\ 0.04 \\ 0.05 \\ -0.02 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.04 \\ 0.05 \\ 0.04 \\ 0.05 \\ 0.$	
Variables	(1) CON_KIM (3) CSR (4) CSRS (5) IO (6) ANA (7) EMP (8) SH (9) ADV (10) PRC (11) TRD (12) ILR (13) CC (14) OM (14) OM (15) CC (15) CC (16) BIC4 (17) AZ (18) CFV (19) DUAL (19) DUAL (20) GROWTH (21) LEV (22) LISTAGE (23) LOSS (24) MB (25) ROA (26) SIGMA (26) SIGMA (27) SIZE (28) SOE (28) SOE	Table 5.

	CON_KIM					
Dependent variable	Model 1	Model 2	Model 3	Model 4		
Intercept	0.0790 (1.80)*	-0.0171 (-0.39)	0.1676 (4.01)***	0.0426 (1.04)		
CSR	0.0020 (8.10)***	0.0014 (6.07)***				
CSRS			0.0255 (8.59)***	0.0178 (6.62)***		
AZ		0.0004 (0.73)		0.0004(0.75)		
CFV		0.1334 (2.52)***		0.1329 (2.52)***		
DUAL		0.0113 (1.99)**		0.0115 (2.03)**		
GROWTH		0.0129 (1.49)		0.0129 (1.49)		
LEV		0.0307 (4.19)***		0.0307 (4.20)***		
LISTAGE		0.0390 (7.94)***		0.0390 (7.96)***		
LOSS		0.0090 (1.15)		0.0090 (1.15)		
MB		-0.0041 (-2.44)***		-0.0041 (-2.43)**		
ROA		0.0479 (0.86)		0.0481 (0.86)		
SIGMA		-0.4114(-2.18)**		-0.4010(-2.14)		
SIZE		0.2122 (6.86)***		0.2097 (6.81)		
SOE		0.0077 (1.40)		0.0076 (1.39)		
TOP (%)		0.0003 (1.88)*		0.0002 (1.83)		
Year fixed effect	Yes	Yes	Yes	Yes		
Industry fixed effect	Yes	Yes	Yes	Yes		
ADJ-RSQ	0.70	0.77	0.70	0.77		

Notes: The table reports the estimated results of the OLS regression. Accounting conservatism measure is *CON_KIM*. Corporate social responsibility score includes *CSR* is the CSR score ranging from 1 (the lowest) to 100 (the highest), and *CSRS* is the standardized CSR score, defined as the difference between CSR and the average value of CSR, scaled by the standard deviation of CSR for firm i at time t. Control variables include Altman's Z-score (*AZ*), volatility of cash flows (*CFV*), CEO duality (*DUAL*), sales revenue growth rate (*GROWTH*), firm's leverage (*LEV*), firm age (*LISTAGE*), relative loss frequency (*LOSS*), market value of equity to book value of equity (*MB*), the return of asset (ROA), volatility (*SIGMA*), firm's size (*SIZE*), stateowned enterprise (*SOE*) and largest shareholder ratios (*TOP* %). The *t*-statistics are reported in parenthesis. ****, ** denote statistical significance at 1%, 5% and 10%, respectively

Table 6.
Baseline result

preference among companies for providing high-quality financial reports after they publish CSR reports (Kim et al., 2012).

4.2 Potential channels through corporate social responsibility influences accounting conservatism

Next, we further explore the potential influence channels of CSR. Referring to Hou and Moskowitz (2005), we examine whether CSR is affected by investor attention and liquidity, and we carry out the regression described by equation (6). Investor attention measures comprise institutional ownership (IO), analyst coverage (ANA), number of employees (EMP), number of shareholders (SH) and advertising expenses (ADV). Liquidity variables include share price (PRC), number of trading days (TRD) and the Amihud (2002) illiquidity ratio (ILR). Our results illustrate that the influence of the proxies for investor attention and liquidity on CSR is mixed, and obvious relationships cannot be determined.

$$\begin{split} \textit{CSR}_{i,t} &= \gamma_0 + \gamma_1 IO_{i,t} + \gamma_2 \textit{ANA}_{i,t} + \gamma_3 \textit{EMP}_{i,t} + \gamma_4 \textit{SH}_{i,t} + \gamma_5 \textit{ADV}_{i,t} + \gamma_6 \textit{PRC}_{i,t} \\ &+ \gamma_7 \textit{TRD}_{i,t} + \gamma_8 \textit{ILR}_{i,t} + \varepsilon_{i,t} \end{split}$$

$$\hat{CSR}_{ATT\ i,t} = \gamma_1 I \hat{O}_{i,t} + \gamma_2 A \hat{NA}_{i,t} + \gamma_3 E \hat{MP}_{i,t} + \gamma_4 S \hat{H}_{i,t} + \gamma_5 A \hat{DV}_{i,t}$$
(7) Accounting conservatism

$$\hat{CSR}_{LIQ\ i,t} = \gamma_6 P\hat{RC}_{i,t} + \gamma_7 T\hat{RD}_{i,t} + \gamma_8 IL\hat{R}_{i,t}$$
(8)

$$CSR_{error\ i,t} = CSR - C\hat{S}R_{ATT\ i,t} - C\hat{S}R_{LIQ\ i,t}$$
 (9)

We separately extract the portion of CSR that is affected by investor attention, the portion of CSR that is affected by liquidity and the portion of CSR that is not explained by investor attention or liquidity. We thereby engage in principal component analysis of the investor attention and liquidity measure variables. The relationship between the investor attention variables and CSR_{ATT} $_{i,t}$ (i.e. the component of attention (ATT) that is determined by CSR) is illustrated in equation (7). The relationship between $C\hat{S}R_{LIQ}$ $_{i,t}$ (i.e., the component of liquidity (LIQ) that is determined by CSR) and the stock liquidity variables is presented in equation (8). The CSR that is not explained by either investor attention or liquidity is CSR_{error} $_{i,t}$, as illustrated in equation (9).

Next, we carry out mixed OLS estimation as presented in equation (10) to explore, which of these two influence channels has a greater effect on accounting conservatism. As illustrated in Table 7, investors pay attention to CSRS as calculated by ATT, and this has a significant positive effect on accounting conservatism. The regression coefficient is 0.005 (0.0756), which is significant at the 1% significance level. However, for LIQ, only the calculated CSRS has a similar effect on accounting conservatism, and the calculated CSR has no obvious influence on accounting conservatism. CSR that is not influenced by investor attention or liquidity results in a significant boost to accounting conservatism, which guides our subsequent research. Therefore, we believe

		CON_KIM				
Dependent variable	Model 1	Model 2	Model 3	Model 4		
Intercept	0.2839 (0.26)	-0.0014 (-0.00)	1.3903 (6.45)***	0.7561 (3.66)***		
CSR_{ATT}	0.0050 (11.98)***	0.0031 (7.50)***				
CSR_{LIQ}	0.0003 (0.14)	0.0003 (0.14)				
CSR_{error}	0.0008 (3.16)***	0.0007 (3.20)***				
$CSRS_{ATT}$			0.0756 (14.84)***	0.0484 (9.00)***		
$CSRS_{LIQ}$			0.0413 (4.55)***	0.0289 (3.39)***		
$CSRS_{error}$			0.0025 (0.75)	0.0025 (0.75)		
Control variables	No	Yes	No	Yes		
Year fixed effect	Yes	Yes	Yes	Yes		
Industry fixed effect	Yes	Yes	Yes	Yes		
ADJ-RSQ	0.72	0.78	0.74	0.78		

Notes: This table reports the pooled OLS regression results of accounting conservatism measures on the investor attention CSR_{ATT} , and stock liquidity determined by information ratings disclosure CSR_{LIQ} . Accounting conservatism measure is CON_KIM . Control variables include Altman's Z-score (AZ), volatility of cash flows (CFV), CEO duality (DUAL), sales revenue growth rate (GROWTH), firm's leverage (LEV), firm age (LISTAGE), relative loss frequency (LOSS), market value of equity to book value of equity (MB), the return of asset (ROA), volatility (SIGMA), firm's size (SIZE), state-owned enterprise (SOE) and largest shareholder ratios (TOP%). ***, ***, * denote statistical significance at the 1%, 5% and 10% levels, based on year and industry fixed effects, respectively

Table 7.
Potential channels
through CSR.
influences
accounting
conservatism

that high CSR scores increase accounting conservatism primarily by attracting investor attention, and that the path of CSRs' influence on improving accounting conservatism through liquidity is not as significant.

To prove that our results are sufficiently robust, we control the regressions of CSCATT and The CSR component of investor liquidity channel (CSRLIQ) for accounting conservatism after controlling the control variables as illustrated in equation (10). The same results are again obtained, which again proves that CSR contributes to increased accounting conservatism in financial reports by attracting investor attention and improving liquidity. The improvement of accounting conservatism is more obvious when its influence is channeled through increased investor attention.

$$CON_{i,t} = \pi_0 + \pi_1 CSR_{ATT\ i,t} + \pi_2 CSR_{LIQ\ i,t} + \pi_3 CSR_{error\ i,t} + \sum_{i,t} \pi_{i,t} CONTROL_{i,t} + \varepsilon_{i,t}$$

$$(10)$$

4.3 Endogeneity

As discussed earlier in this paper, stronger corporate governance leads to improved information disclosure (Bae et al., 2006), and companies with high levels of accounting conservatism are more likely to have high-quality CSR disclosures than companies with low levels of accounting conservatism. Prior studies have also revealed that CSR is a product of financial performance (Hong et al., 2012). Therefore, CSR is likely to be endogenous. To increase the credibility of our research, we refer to Larcker and Rusticus (2010) for a discussion of disclosure issues, and we follow their endogenous problem-solving steps. First, six instrumental variables are chosen to test the possible endogeneity of CSR, namely, log shareholder (LOGSH), sales growth (SG), capital intensity (CI), operating margin (OM), operating cycle (OC) and the Big-four auditor dummy variable (BIG4). The results of each test are exhibited in Panel A, Table 8. We perform the Durbin-Wu-Hausman test for the existence of endogeneity, which reveals that the explanatory variable CSR is endogenous. Therefore, when studying the relationship between accounting conservatism and CSR, we must focus on the effect of endogeneity. Next, we perform an overidentification test for these six instrumental variables, and finally, select CI, OM, OC and BIG4 as the instrumental variables of CSR. We then test for indications of problems related to weak instrumental variables, the results of which are exhibited in Panel A of Table 8. We then apply the twostage least squares equations (11) and (12) to control exogenous control variables, solve endogeneity problems and estimate the relationship between CSR and accounting conservatism.

$$CSR_{i,t} = \epsilon_1 + \epsilon_2 CI_{i,t} + \epsilon_3 OM_{i,t} + \epsilon_4 OC_{i,t} + \epsilon_5 BIG4_{i,t} + \sum \epsilon_{i,t} CONTROL_{i,t} + \epsilon_{i,t}$$
(11)

$$CON_{i,t} = \varnothing_0 + \varnothing_1 CSR_{2SLS\ i,t} + \sum \varnothing_{i,t} CONTROL_{i,t} + \varepsilon_{i,t}$$
(12)

The results of the two-stage least squares regression are exhibited in Panel B of Table 8. The results of the first stage regression reveal that the relationships between the two instrumental variables and OC, BIG4 and CSR are significant. The results of the second phase regression reveal that an increase in CSR disclosure rating can significantly increase

Accounting
conservatism

CSR

5.66

(0.02)

5.76

(0.02)

4.06 (0.26)

4.06 (0.26)

Kleibergen-Paap W	ald F statistic			11.95	
Panel B: 2SLS					
Dependent variable	First stage: CSR		Second : CON_	KIM	
		Model 1	Model 2	Model 3	Model 4
	-22.4589 (-2.38)***	-0.1329(-2.19)**	0.0192 (0.66)	23.2131 (1.81)*	2.2692 (2.61)***
CI	0.2749 (0.13)				
OM	0.8108 (0.32)				
OC	-0.9927 (-2.13)**				
BIG4	4.7430 (4.30)***	0.000= (0.05)****			
CSR_{2SLS}		0.0035 (3.67)***	0.0000 (0.05)***		
$CSRS_{2SLS}$			0.0388 (3.85)***	0.0024 (1.47)	
CSR _{ATT, 2SLS}				0.0034 (1.47)	
CSR _{LIQ, 2SLS}				0.0418 (1.77)*	0.0419 (2.01)**
CSRS _{ATT, 2SLS} CSRS _{LIQ, 2SLS}					0.0987 (2.15)**
Control variables	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes	Yes	Yes
ADJ-RSQ	0.35	165	163	163	103
5	0.00				
Panel C: IV-GMM					
Dependent variable			CON_		
		Model 1	Model 2	Model 3	Model 4
Intercept		-0.1338 (-2.21)**	0.0135 (0.47)	25.7402 (2.02)**	2.2553 (2.60)***
CSR_{2SLS}		0.0034 (3.57)***	0.0075 (0.75***		
$CSRS_{2SLS}$			0.0375 (3.75)***	0.0000 (1.07)	
CSR _{ATT, 2SLS}				0.0029 (1.27)	
CSR _{LIQ, 2SLS}				0.0466 (1.98)**	0.0261 (1.75)*
CSRS _{ATT, 2SLS}					0.0361 (1.75)* 0.1001 (2.18)**
CSRS _{LIQ, 2SLS} Control variables		Yes	Yes	Yes	Yes
Year fixed effect		Yes	Yes	Yes	Yes
Industry fixed effect		Yes	Yes	Yes	Yes
ADJ-RSQ		163	103	165	105

Notes: The table reports the two-stage least squares (2SLS) regression analysis results for examining

whether CSR explains earnings management from 2010 to 2018. Panel A reports endogeneity test results for

CSR. Panel B reports two-stage least squares (2SLS) regression and GMM regression analysis results in

Panel C. CSR as the endogenous variable and overidentifying restrictions test result for choosing the

instrumental variables such as capital intensity (CI), capital intensity (CI), operating margin (OM), operating cycle (OC) and Big-four auditor dummy (BIG4). ***, **, * denote statistical significance at the

1%, 5% and 10% levels, based on year and industry fixed effects, respectively

Panel A

b-value

p-value

p-value

Durbin scores chi^2 (1)

Wu-Hausman F(1,896)

Hansen J statistic

Sargan score chi^2 (3)

Test of weak instruments

Durbin-Wu-Hausman test for endogeneity

Test of over identifying restrictions

Table 8.

Two-stage least squares (2SLS) regression analysis for the relationship between CSR and accounting conservatism accounting conservatism. However, the results of the second stage regression also illustrate that the relationship between CSR and liquidity is more significant than that of CSR and investor attention. Panel C exhibits the second-stage regression, which applies Generalized method of moments (GMM) and the two-stage CSRS and has a regression coefficient of 0.0034~(0.0375) and p-value of less than 0.01. This is statistically significant, indicating that CSRs' influence on accounting conservatism is robust. Applying the same regression analysis on the two-stage CSR, we discover that the liquidity channel results in a larger improvement in accounting conservatism than investor attention, which replicates the results of Panel B. All these results reveal that CSR effectively increases accounting conservatism after selecting appropriate instrumental variables to control the endogeneity problem.

4.4 Robustness check for alternate proxy of accounting conservatism

In the previous section, we discuss another measure of accounting conservatism: the original Khan and Watts (2009) firm-year level of accounting conservatism (CScore). We apply this measure to estimate accounting conservatism under the regression as a robustness test. The results are exhibited in Table 9. The most basic regression results illustrate that CSR can significantly improve accounting conservatism after controlling the year and industry. The two-stage regression suggests that CSR and CSRS have a significant positive influence on accounting conservatism, with regression coefficients of 0.0811 and 0.0799, respectively. The effect of CSR on the path of attracting investors to improve corporate accounting conservatism is also significant. After controlling the exogenous control variables, we obtain the same conclusion. This indicates that our research is robust and that CSR can indeed improve accounting conservatism by attracting investor attention.

4.5 Regression of accounting conservatism using component corporate social responsibility We also investigate the effect of other CSR agents on accounting conservatism, mainly by using four dimensions of CSR in the RKS database as proxies for CSR (Gong and Ho, 2018a, Gong et al., 2019). CSR M refers to the macrocosm dimension, CSR C stands for the content

	CON_KHAN					
Dependent variable	Model 1	Model 2	Model 3	Model 4		
Intercept CSR	-3.2597 (-12.24)*** 0.0663 (10.00)***	-1.1813 (-1.68)* 0.0618 (8.16)***	-3.7273 (-11.89)***	-2.0369 (-2.82)***		
CSR_{2SLS}	0.0003 (10.00)	0.0018 (8.10)	0.0811 (8.99)***	0.0829 (8.47)***		
$CSRS_{2SLS}$ $CSR_{ATT, 2SLS}$			0.0799 (9.78)*** 0.0797 (9.84)***	0.0786 (8.67)*** 0.0782 (8.62)***		
Control variables	NO	YES	NO	YES		
Year fixed effect	YES	YES	YES	YES		
Industry fixed effect	YES	YES	YES	YES		
ADJ-RSQ	0.09	0.13	0.09	0.17		

Table 9.Robustness check for the alternate proxy of accounting conservatism

Notes: This table reports the estimated results of the OLS regression. Control variables include Altman's Z-score (AZ), volatility of cash flows (CFV), CEO duality (DUAL), sales revenue growth rate (GROWTH), firm's leverage (LEV), firm age (LISTAGE), relative loss frequency (LOSS), market value of equity to book value of equity (MB), the return of asset (ROA), volatility (SIGMA), firm's size (SIZE), state-owned enterprise (SOE) and largest shareholder ratios (TOP%). T-statistics is reported in parentheses. ****, ***, ** denote statistical significance at the 1%, 5% and 10% levels, based on year and industry fixed effects, respectively

Accounting

dimension, CSR_T stands for the technical dimension and CSR_I stands for the industry index dimension. After controlling the year, industry and control variables, a mixed OLS regression of the agent variables for each dimension is performed on accounting conservatism and the results are illustrated in Table 10. We discover that increased CSR disclosure for each dimension significantly increases accounting conservatism while other conditions remain unchanged. However, when the four-dimensional CSR is run through the same regression equation, the results reveal that the content dimension and the industry index dimension are more critical in their effect on enterprise accounting conservatism than the other dimensions, and this effect is positive. In general, CSRs' effect on improved accounting conservatism is robust.

4.6 Regression of accounting conservatism using alternate proxy of corporate social responsibility

In addition to using the CSR variables from the RKS database, we examine the results of the professional evaluation system of the CSR report from the Hexun.com CSR database. This is another agent of CSR that has been examined in various prior studies (Wang et al., 2019). The professional evaluation system of listed companies' CSR reports comprises five measures, namely, shareholder responsibility: employee responsibility; supplier, customer and consumer rights responsibility; environmental responsibility; and social responsibility. Each measure has also established secondary and tertiary indicators to conduct a comprehensive evaluation of social responsibility. These indicators comprise 13 secondary indicators and 37 tertiary indicators. A single company report requires the evaluation of at least three experts who have no association with the company. Therefore, the evaluation results are comprehensive, independent and objective.

We choose data for 2010–2018 and perform a regression analysis of the effect of CSR on accounting conservatism. The results are revealed in Table 11. CSR_A is the total score of Hexun CSR, CSR_B is the standardization of the total CSR score and CSR_C is the annual ranking of the Hexun CSR score. Companies at the top of Table 11 have higher CSR scores. The results reveal that the regression coefficients of all CSR measures are positive and the

Dependent variable	Model 1	Model 2	CON_KIM Model 3	Model 4	Model 5
Intercept	-0.0707 (-0.91)	-0.0983(-1.26)	-0.0660(-0.85)	-0.0408 (-0.53)	-0.0833(-1.07)
CSR_M	0.0035 (4.44)***				0.0003 (0.16)
CSR_C		0.0031 (5.19)***			0.0023 (1.98)**
CSR_T			0.0078 (4.16)***		-0.0019(-0.52)
CSR_I				0.0119 (5.46)***	0.0084 (3.24)***
Control variables	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes	Yes	Yes
ADJ-RSQ	0.57	0.58	0.57	0.58	0.58

Notes: This table reports the pooled OLS regression results of alternate proxy of CSR. There are four dimensions of CSR: macrocosm dimension (CSR_M), content dimension (CSR_C), technique dimension (CSR_T) and industry index dimension (CSR_I). Control variables include Altman's Z-score (AZ), volatility of cash flows (CFV), CEO duality (DUAL), sales revenue growth rate (GROWTH), firm's leverage (LEV), firm age (LISTAGE), relative loss frequency (LOSS), market value of equity to book value of equity (MB), the return of asset (ROA), volatility (SIGMA), firm's size (SIZE), state-owned enterprise (SOE) and largest shareholder ratios (TOP %). ****, **, ** denote statistical significance at the 1%, 5% and 10% levels, based on year and industry fixed effects, respectively

Table 10.
Regression of accounting conservatism using component CSR

Dependent variable	CON_KIM		
	Model 1	Model 2	Model 3
Panel A: Main results			
Intercept	0.0317 (0.71)	0.0606 (1.36)	0.0984 (2.19)**
CSR_A	0.0012 (9.66)***	0.0102 (0.01) kakak	
CSR_B CSR C		0.0193 (9.01)***	0.0000 (-8.01)***
Control variables	No	No	No
Year fixed effect	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes
ADJ-RSQ	0.09	0.09	0.08
Panel B: After controlling co			
Intercept CSR A	0.0123 (0.32) 0.0008 (6.92)***	0.0307 (0.79)	0.0567 (1.44)
CSR_B	, ,	0.0125 (6.22)***	
CSR_C			0.0000 (-5.99)***
Control variables	Yes	Yes	Yes
Year fixed effect Industry fixed effect	Yes Yes	Yes Yes	Yes Yes
ADJ-RSQ	0.40	0.40	0.40

Tabel 11. Regression of accounting conservatism using the alternate proxy of CSR

Notes: This table reports the pooled OLS regression results of alternate proxy of CSR. CSR_A is the total score of Hexun CSR, CSR_B is the standardization of CSR total score, CSR_C is the annual ranking of Hexun CSR score and the CSR score is higher on the top. Control variables include Altman's Z-score (AZ), volatility of cash flows (CFV), CEO duality (DUAL), sales revenue growth rate (GROWTH), firm's leverage (LEV), firm age (LISTAGE), relative loss frequency (LOSS), market value of equity to book value of equity (MB), the return of asset (ROA), volatility (SIGMA), firm's size (SIZE), state-owned enterprise (SOE) and largest shareholder ratios (TOP%). ****, **, ** denote statistical significance at the 1%, 5% and 10% levels, based on year and industry fixed effects, respectively

results remain significant after controlling the control variables. Enterprises with higher CSR scores have more stable accounting policies than firms with lower CSR scores. In other words, CSR can effectively improve the accounting conservatism of enterprises. This result is consistent with our previous findings.

Next, we further explore the robustness of the channels of CSRs' effect on accounting conservatism. The three CSR indicators of the Hexun.com CSR database are used to extract the portions affected by liquidity and investor attention, and the two parts are subtracted from the original CSR value. The difference is the portion of CSR that is not explained by investor attention or liquidity. A regression of CON_KIM is conducted to explore the significant robustness of the influence channels, and the results are illustrated in Table 12.

The results presented in Table 12 support our argument that liquidity and investor attention are the main channels by which CSR affects accounting conservatism. This result is more pronounced after controlling the control variables, as the total CSR scores that are determined by investor attention and liquidity have regression coefficients of 0.0034 and 0.0033, respectively, for CON_KIM, which are both statistically significant. The regression coefficients of CON_KIM, which are influenced by investor attention and liquidity, are 0.0593 and 0.0550, respectively, which are significant at the 1% significance level. Therefore, both investor attention and liquidity are the main channels by which CSR affects accounting conservatism. Moreover, the larger the CSR score is, the higher the annual ranking of the total CSR score is. Therefore, the negative regression coefficient indicates that CSR has the

Dependent variable	Model 1	CON_KIM Model 2	Model 3	Accounting conservatism
Panel A: Main results Intercept	-0.1425 (-3.15)***	-0.1214 (-2.32)**	-0.1838 (-2.66)***	
CSR_A _{ATT}	0.0051 (15.24)***	-0.1214 (-2.52)	-0.1030 (-2.00)***	
CSR_A _{LIQ} CSR_A _{error}	0.0014 (2.13)** 0.0002 (1.21)			
CSR_B_{ATT}	0.0002 (1.21)	0.0928 (16.32)***		
CSR_B _{LIQ} CSR_B _{error}		0.0165 (1.66)* 0.0016 (0.63)		
CSR_C _{ATT} CSR_C _{LIQ} CSR_C _{error}			-0.0001 (-13.15)*** 3.E-06 (0.29) -6.E-06 (-1.99)**	
Control variables Year fixed effect	No Yes	No Yes	No Yes	
Industry fixed effect	Yes	Yes	Yes	
ADJ-RŠQ	0.13	0.13	0.11	
Panel B: After controlling				
Intercept CSR_A _{ATT} CSR_A _{LIQ} CSR_A _{error}	-0.1070 (-2.71)*** 0.0034 (11.40)*** 0.0033 (5.78)*** 0.0002 (1.60)	0.0308 (0.66)	0.0320 (0.53)	
CSR_B _{ATT} CSR_B _{LIQ} CSR_B _{error}	0.0002 (1.00)	0.0593 (11.71)*** 0.0550 (6.06)*** 0.0020 (0.94)		
CSR_C _{ATT} CSR_C _{LIQ} CSR_C _{error}			-0.0001 (-10.25)*** -3.E-05 (-3.88)*** -5.E-06 (-2.05)**	Table 12. The robustness of the
Control variables	Yes	Yes	Yes	channels of alternate
Year fixed effect	Yes	Yes	Yes	CSRs' effect on
Industry fixed effect ADJ-RSQ	Yes 0.42	Yes 0.42	Yes 0.41	accounting conservatism
III) IWQ	0.12	0.12	0.11	content rations

effect of improving accounting conservatism. These results are consistent with our previous research and provide increased persuasive power to prove that our findings are robust.

5. Discussion and implications

5.1 Practical implications

Our research confirms the importance of non-financial CSR information disclosure for accounting policy choices. A critical part of our research is its examination of the effect of investor attention and liquidity, which confirms that CSR disclosure can improve accounting conservatism through these two market reaction channels. CSR disclosure in China started relatively recently; therefore, our research contributes to CSR disclosure standardization and the understanding of the effects of CSR information disclosure. Conditional accounting conservatism limits managers' ability to exaggerate performance and hide negative news, thus providing investors with an improved information environment than would otherwise be the case (Kim and Zhang, 2016). Therefore, we explore factors that affect accounting conservatism, which is also critical for investor protection.

CSR disclosure can illustrate the corporate accounting conservatism that is difficult for investors to observe directly. It protects investors by providing a means of measurement

and understanding of corporate information flow and disclosure. The prevalence of CSR reports in China has increased significantly; however, the quality of CSR reports is still relatively low. Therefore, the external supervision of the CSR disclosure system needs to be further improved. In terms of laws and regulations, it is also necessary to strengthen supervision and formulate comprehensive and specific laws and regulations; to standardize disclosure deadlines, methods, procedures and content of CSR reports; and to expand the scope of compulsory disclosure by enterprises. In addition, it is necessary to guide companies to complete third-party verification of CSR reports and to standardize the verification workflow to improve reports' quality and credibility. When formulating laws and regulations and managing enterprises, relevant regulatory agencies should maintain consistency of regulatory standards and should unify the scale of law enforcement.

Our research reveals that investor attention is one of the main channels through which CSR affects accounting conservatism. Companies with outstanding CSR performance attract specialized institutional investors and analyst reports (Dhaliwal et al., 2011). As critical members of the investment community, analysts can obtain and interpret information more effectively than other investors can through their capabilities; they can thus, transmit information to the market, which reduces information opacity and helps investors obtain and understand company-specific information. Increased analyst attention can attract increased investment (Tsao et al., 2016). However, analyst attention may introduce other problems such as conflicts of interest and excessive optimism, which can have negative effects for attracting investors. Therefore, strengthening the supervision and management of analysts is crucial. To improve the objectivity and comparability of analysis and interpretation and to improve the accuracy of analyst forecasts, regulators should pay attention to the training of analysts' abilities and moral qualities. Industry associations also need to strengthen self-regulation to play a greater role in regulating analyst behavior than they do at present. In addition, institutional investors can provide external investors with information on company characteristics through their trading behavior; therefore, they can also increase information transparency. However, institutional investors sometimes engage in irrational behaviors such as herding and holding shares. Therefore, strengthening education and guidance to institutional investors is another critical component of external supervision. Government authorities should encourage institutional investors to embrace the concept of long-term investment, to enhance their ability to collect and analyze companyspecific information including CSR reports, and to prevent insider information-based transactions. Doing so would effectively protect investor interests and would further promote the influence of CSR reports.

From the perspective of a company's internal decision-making and governance, a CSR report discloses CSR activities to corporate stakeholders and provides them with information on corporate financial decisions. Because of widespread agency problems arising from the separation of ownership and management in modern enterprises, managers may engage in self-interested behavior by using CSR reports to hide information or to disclose false information, thus distorting investors' understanding of financial policies. Therefore, companies must strengthen their corporate governance and establish a sound CSR information disclosure monitoring mechanism. This is a protective action for investors that can also establish a positive reputation for an enterprise, and thus, contribute to its long-term development.

5.2 Managerial insight

Managers of different corporations have very different attitudes toward CSR. Some managers who understand the effect of CSR disclosure already disclose CSR information actively. However,

motivations for CSR disclosure are very different among different managers. Some hope to use CSR to conceal unethical behavior at their firms. Others focus on promoting their firms' corporate image through CSR by establishing and maintaining a positive corporate reputation, reducing information opacity and providing investors with transparent information (Cui et al., 2015). By contrast, some managers do not understand the role of CSR disclosure, and thus, do not focus on CSR activities or relevant disclosure. Based on a sample of Chinese listed companies, we discover that companies with different CSR disclosure characteristics exhibit significant differences in their financial reporting policies. CSR can significantly increase accounting conservatism. In addition, CSR can provide outside investors with increased indirect information regarding standards for financial statement preparation, which fosters an environment of improved investor protection, attracts additional investor attention and improves the liquidity of listed securities. Managers of enterprises that have not adopted CSR information disclosure should pay attention to it and should include the disclosure of CSR reports as part of the enterprise information disclosure at their firms. A company that discloses its CSR activities through CSR reports can improve its reputation, build an environment of increased investor protection and increase investor trust. This effectively reduces the company's risks and benefits its future development. For example, Lins et al. (2017) revealed that during financial crises, the trust between listed companies and investors and other stakeholders that is established by corporate participation in CSR activities is enhanced. Listed companies that engage in extensive CSR disclosure have higher stock returns, profitability and business growth than firms with limited CSR disclosure. Managers of companies that have disclosed CSR information should also increase their emphasis on CSR reports, including the consideration of separate and more frequent disclosure of CSR reports than they currently provide, instead of viewing them as ancillary reports of their corporate financial statements. Management should also consciously improve the quality of disclosure and provide authentic, accurate and effective information to external investors to maximize the effect of CSR disclosure.

5.3 Limitations of the study

Our research analyzes the effect of CSR on accounting conservatism, but a consistent definition of CSR has not been established. Although the main objectives of the various definitions in the relevant literature have substantial similarities, they differ in the breadth of their connotations. Therefore, a unified measure of CSR does not exist (Sheehy, 2015). Based on the relevant literature, we use the RKS CSR report score and CSR report indicator from Hexun.com to measure the CSR disclosure quality. Although the data from these evaluation systems support our results, further improvements on the CSR measurement method are necessary to ensure that the CSR indicators can evaluate the quality of CSR reports more objectively, comprehensively and thoroughly than is possible in this study. In particular, research on the subdivision dimensions of the CSR indicators could improve our understanding of the role and influence of various stakeholders, and thus, improve our understanding of the effects of CSR as non-financial information on accounting conservatism

Our results are based on empirical research that examines real data, and this study does not discuss theoretical models. We provide a seemingly reasonable explanation for the results of our empirical analysis. Because numerous market factors are interrelated, the social environment is a critical determinant of the disclosure and post-disclosure effects of CSR (Dhaliwal et al., 2014). Some control variables such as the nature of a company, its governance structure and its business performance, are also crucial. Therefore, more complex models or empirical methods than those that we use in this study might provide further insights than those

of our study. In addition, the introduction of a theoretical derivation to support our research can provide more convincing results than those of our study.

5.4 Future research

Our research confirms the importance of CSR disclosure for corporate financial policy choices. Companies with high-quality CSR disclosure tend to exhibit a higher degree of accounting conservatism than other companies do. Because the RKS database and data from Hexun.com include professional evaluations of CSR reports, our research scope comprises companies that have disclosed CSR reports; it does not include companies that have not disclosed CSR information. Future research might consider analyzing 0 and 1 variables (i.e. whether a company discloses a CSR report) to examine differences in the effect on accounting conservatism between the two types of companies and the reasons for those differences.

We also discover that CSR improves accounting conservatism through increased securities liquidity and investor attention. As for whether other channels of CSR disclosures' effect of increasing accounting conservatism exist, our research results are inconclusive. Further research is required to explore whether additional transmission paths exist for the influence of CSR information on accounting conservatism.

With the rapid development of information technology in recent years, electronic disclosure is gradually becoming widespread. This new disclosure mode is very different from traditional information disclosure in terms of its content and method. This raises questions regarding whether electronic disclosure affects CSR or accounting conservatism and whether the relationship between CSR and accounting conservatism remains robust in this mode of disclosure. These topics are worthy of future examination.

The agent variables of market reaction channels have specific characteristics and can also have different effects, which requires further detailed analysis. For example, when considering the role of investor attention, analysts' independence, affiliation with underwriters and ranking may all be affected by differences in CSR disclosure, and thus, may have different effects on accounting conservatism. In addition, the various types of institutional investors exhibit specific characteristics and differences in their investment motivation and supervisory participation (An and Zhang, 2013), which may also result in different effects on accounting conservatism. Future research should consider extending on this study's static analysis to incorporate dynamic comparative research. This study's methodology can also be applied to data from different information disclosure environments and social backgrounds in other countries or regions to compare the results.

6. Conclusion

We investigate the relationship between CSR ratings and accounting conservatism. We examine data on companies listed on China's SHSE and SZSE from 2010 to 2018, which confirms the results of previous research: companies with higher CSR have higher accounting conservatism than companies with lower CSR, which implies that companies with high-quality information disclosure are more inclined to adopt conservative accounting policies to reduce information asymmetry between internal managers and external stakeholders than companies with low-quality information disclosure. Our research reveals that companies with high CSR disclosure scores can attract investor attention and analyst coverage, which is the main channel by which CSR promotes accounting conservatism. After controlling other variables that may affect accounting conservatism, firm-specific characteristics and some variables with Chinese characteristics, the same result is obtained: the promotion of CSR improves corporate accounting conservatism. We conduct detailed

endogeneity testing and robustness testing, and the positive relationship between CSR and accounting conservatism remains significant.

One of this papers' innovations is to supplement previous research on the relationship between information disclosure and financial information quality. For the first time, we investigate the effect of CSR disclosure, which is non-financial information, on accounting conservatism and the channels of CSRs' influence on accounting conservatism, which expands on the relevant literature. Our results reveal that increased analyst and investor attention toward companies with high CSR information quality is the primary factor that drives companies to choose conservative accounting methods, rather than increased stock liquidity. Our results also provide possible direction for subsequent research. The results of the examination of the channels of CSRs' effect on accounting conservatism indicate the existence of other paths by which CSR significantly affects accounting conservatism. These other influence paths merit further exploration.

Accounting conservatism

References

- Ahmed, A.S. and Duellman, S. (2013), "Managerial overconfidence and accounting conservatism", *Journal of Accounting Research*, Vol. 51 No. 1, pp. 1-30.
- Ahmed, A.S., Billings, B.K., Morton, R.M. and Stanford-Harris, M. (2002), "The role of accounting conservatism in mitigating bondholder-shareholder conflicts over dividend policy and in reducing debt costs", *The Accounting Review*, Vol. 77 No. 4, pp. 867-890.
- Albuquerque, R., Koskinen, Y. and Zhang, C. (2019), "Corporate social responsibility and firm risk: theory and empirical evidence", *Management Science*, Vol. 65 No. 10, pp. 4451-4469.
- Amihud, Y. (2002), "Illiquidity and stock returns: cross-section and time-series effects", *Journal of Financial Markets*, Vol. 5 No. 1, pp. 31-56.
- An, H. and Zhang, T. (2013), "Stock price synchronicity, crash risk, and institutional investors", *Journal of Corporate Finance*, Vol. 21, pp. 1-15.
- An, H., Hardin, W. and Wu, Z. (2012), "Information asymmetry and corporate liquidity management: evidence from real estate investment trusts", The Journal of Real Estate Finance and Economics, Vol. 45 No. 3, pp. 678-704.
- Armstrong, C.S., Core, J.E., Taylor, D.J. and Verrecchia, R.E. (2011), "When does information asymmetry affect the cost of capital?", *Journal of Accounting Research*, Vol. 49 No. 1, pp. 1-40.
- Aupperle, K.E., Carroll, A.B. and Hatfield, J.D. (1985), "An empirical examination of the relationship between corporate social responsibility and profitability", *Academy of Management Journal*, Vol. 28 No. 2, pp. 446-463.
- Bae, K., Lim, C. and Wei, J. (2006), "Corporate governance and conditional skewness in the world's stock markets", *The Journal of Business*, Vol. 79 No. 6, pp. 2999-3028.
- Ball, R., Robin, A. and Sadka, G. (2008a), "Is financial reporting quality shaped by equity markets or by debt markets? An international study of timeliness and conservatism", *Review of Accounting Studies*, Vol. 13 Nos 2/3, pp. 168-205.
- Ball, R., Bushman, R.M. and Vasvari, F.P. (2008b), "The debt-contracting value of accounting information and loan syndicate structure", *Journal of Accounting Research*, Vol. 46 No. 2, pp. 247-287.
- Basu, S. (1997), "The conservatism principle and the asymmetric timeliness of earnings", *Journal of Accounting and Economics*, Vol. 24 No. 1, pp. 3-37.
- Becchetti, L., Ciciretti, R., Hasan, I. and Kobeissi, N. (2012), "Corporate social responsibility and shareholder's value", *Journal of Business Research*, Vol. 65 No. 11, pp. 1628-1635.
- Benlemlih, M. and Bitar, M. (2018), "Corporate social responsibility and investment efficiency", *Journal of Business Ethics*, Vol. 148 No. 3, pp. 647-671.

- Bereskin, F., Byun, S.K., Officer, M.S. and Oh, J.-M. (2018), "The effect of cultural similarity on mergers and acquisitions: evidence from corporate social responsibility", *Journal of Financial and Quantitative Analysis*, Vol. 53 No. 5, pp. 1995-2039.
- Beyer, A., Cohen, D., Thomas, Z. and Walther, B. (2010), "The financial reporting environment: review of the recent literature", *Journal of Accounting and Economics*, Vol. 50 Nos 2/3, pp. 296-343.
- Bhandari, A. and Javakhadze, D. (2017), "Corporate social responsibility and capital allocation efficiency", *Journal of Corporate Finance*, Vol. 43, pp. 354-377.
- Boone, A. and White, J.T. (2015), "The effect of institutional ownership on firm transparency and information production", *Journal of Financial Economics*, Vol. 117 No. 3, pp. 508-533.
- Bozzolan, S., Fabrizi, M., Mallin, C.A. and Michelon, G. (2015), "Corporate social responsibility and earnings quality: international evidence", *The International Journal of Accounting*, Vol. 50 No. 4, pp. 361-396.
- Brennan, M.J. and Tamarowski, C. (2000), "Investor relations, liquidity, and stock prices", *Journal of Applied Corporate Finance*, Vol. 12 No. 4, pp. 26-37.
- Bushman, R.M. and Smith, A.J. (2001), "Financial accounting information and corporate governance", *Journal of Accounting and Economics*, Vol. 32 Nos 1/3, pp. 237-333.
- Byun, S.K. and Oh, J.M. (2018), "Local corporate social responsibility, media coverage, and shareholder value", *Journal of Banking and Finance*, Vol. 87, pp. 68-86.
- Callen, J.L., Khan, M. and Lu, H. (2013), "Accounting quality, stock price delay, and future stock returns", Contemporary Accounting Research, Vol. 30 No. 1, pp. 269-295.
- Chen, Y., Hung, M.Y. and Wang, Y. (2018), "The effect of mandatory CSR disclosure on firm profitability and social externalities: evidence from China", *Journal of Accounting and Economics*, Vol. 65 No. 1, pp. 169-190.
- Chen, C.L., Liu, Q., Li, J. and Wang, L.F.S. (2016), "Corporate social responsibility and downstream price competition with retailer's effort", *International Review of Economics and Finance*, Vol. 46, pp. 36-54.
- Cheng, C.-L. and Kung, F.-H. (2016), "The effects of mandatory corporate social responsibility policy on accounting conservatism", *Review of Accounting and Finance*, Vol. 15 No. 1, pp. 2-20.
- Cheung, Y.L., Tan, W. and Wang, W. (2018), "National stakeholder orientation, corporate social responsibility, and bank loan cost", *Journal of Business Ethics*, Vol. 150 No. 2, pp. 505-524.
- Chih, H.L., Shen, C.H. and Kang, F.C. (2008), "Corporate social responsibility, investor protection, and earnings management: some international evidence", *Journal of Business Ethics*, Vol. 79 Nos 1/2, pp. 179-198.
- Chung, H., Elder, J. and Kim, J.C. (2010), "Corporate governance and liquidity", *Journal of Financial and Quantitative Analysis*, Vol. 45 No. 2, pp. 265-291.
- Chung, H., Judge, W.Q. and Li, Y.H. (2015), "Voluntary disclosure, excess executive compensation, and firm value", *Journal of Corporate Finance*, Vol. 32, pp. 64-90.
- Core, J.E. (2001), "A review of the empirical disclosure literature: discussion", *Journal of Accounting and Economics*, Vol. 31 Nos 1/3, pp. 441-456.
- Cui, J., Jo, H. and Na, H. (2015), "Does corporate social responsibility affect information asymmetry?", Journal of Business Ethics, Vol. 148 No. 3, pp. 549-572.
- Dhaliwal, D., Li, O.Z., Tsang, A. and Yang, Y.G. (2011), "Voluntary nonfinancial disclosure and the cost of equity capital: the initiation of corporate social responsibility reporting", *The Accounting Review*, Vol. 86 No. 1, pp. 59-100.
- Dhaliwal, D., Li, O.Z., Tsang, A. and Yang, Y.G. (2014), "Corporate social responsibility disclosure and the cost of equity capital: the roles of stakeholder orientation and financial transparency", *Journal of Accounting and Public Policy*, Vol. 33 No. 4, pp. 328-355.
- Diamond, D.W. and Verrecchia, R.E. (1991), "Disclosure, liquidity, and the cost of capital", *The Journal of Finance*, Vol. 46 No. 4, pp. 1325-1359.
- Dolores Sánchez-Fernández, M., Vargas-Sánchez, A. and Remoaldo, P. (2014), "Institutional context and hotel social responsibility", *Kybernetes*, Vol. 43 Nos 3/4, pp. 413-426.

Accounting

- Dyck, A., Lins, K.V., Roth, L. and Wagner, H.F. (2019), "Do institutional investors drive corporate social responsibility? International evidence", *Journal of Financial Economics*, Vol. 131 No. 3, pp. 693-714.
- Ee, M.S., Chao, C.C., Wang, L.F.S. and Eden, S.H. (2018), "Environmental corporate social responsibility, firm dynamics and wage inequality", *International Review of Economics and Finance*, Vol. 56, pp. 63-74.
- El Ghoul, S., Guedhami, O., Kwok, C.C. and Mishra, D.R. (2011), "Does corporate social responsibility affect the cost of capital?", *Journal of Banking and Finance*, Vol. 35 No. 9, pp. 2388-2406.
- El Ghoul, S., Guedhami, O., Wang, H. and Kwok, C.C. (2016), "Family control and corporate social responsibility", *Journal of Banking and Finance*, Vol. 73, pp. 131-146.
- Francis, J.R. and Wang, D. (2008), "The joint effect of investor protection and big 4 audits on earnings quality around the world", *Contemborary Accounting Research*, Vol. 25 No. 1, pp. 157-191.
- Francis, J., Nanda, D. and Olsson, P. (2008), "Voluntary disclosure, earnings quality, and cost of capital", *Journal of Accounting Research*, Vol. 46 No. 1, pp. 53-99.
- French, D., McKillop, D. and Stewart, E. (2019), "The effectiveness of smartphone apps in improving financial capability", *The European Journal of Finance*, Vol. 26 Nos 4/5, pp. 1-17.
- Georgeta, S., Anca, M. and Sorin, P. (2016), "Role of CSR on the digital single market from the standpoint of Romania", Annals of the University of Petrosani Economics, Vol. 16, pp. 253-264.
- Gigler, F., Kanodia, C., Sapra, H. and Venugopalan, R. (2009), "Accounting conservatism and the efficiency of debt contracts", *Journal of Accounting Research*, Vol. 47 No. 3, pp. 767-797.
- Goldfarb, A. and Tucker, C. (2019), "Digital economics", Journal of Economic Literature, Vol. 57 No. 1, pp. 3-43.
- Gong, Y. and Ho, K.C. (2018a), "Does corporate social responsibility matter for corporate stability? Evidence from China", *Quality and Quantity*, Vol. 52 No. 5, pp. 2291-2319.
- Gong, Y. and Ho, K.C. (2018b), "Corporate social responsibility and managerial short-termism", *Asia-Pacific Journal of Accounting and Economics*, Vol. 38, pp. 354-373.
- Gong, G., Xu, S. and Gong, X. (2018), "On the value of corporate social responsibility disclosure: an empirical investigation of corporate bond issues in China", *Journal of Business Ethics*, Vol. 150 No. 1, pp. 227-258.
- Gong, Y., Ho, K.C., Lo, C.C., Karathanasopoulos, A. and Jiang, I.M. (2019), "Forecasting price delay and future stock returns: the role of corporate social responsibility", *Journal of Forecasting*, Vol. 38 No. 4, pp. 354-373.
- Guay, W. and Verrecchia, R. (2006), "Discussion of an economic framework for conservative accounting and bushman and piotroski", *Journal of Accounting and Economics*, Vol. 42 Nos 1/2, pp. 149-165.
- Hail, L. (2002), "The impact of voluntary corporate disclosure on the ex-ante cost of capital for swiss firms", European Accounting Review, Vol. 11 No. 4, pp. 741-773.
- Healy, P. and Palepu, K.G. (2001), "Information asymmetry, corporate disclosure, and the capital market: a review of the empirical disclosure literature", *Journal of Accounting and Economics*, Vol. 31 Nos 1/3, pp. 405-440.
- Healy, P., Hutton, A. and Palepu, K. (1999), "Stock performance and intermediation changes surrounding sustained increases in disclosure", Contemporary Accounting Research, Vol. 16 No. 3, pp. 485-520.
- Heflin, F.L., Shaw, K.W. and Wild, J.J. (2005), "Disclosure policy and market liquidity: impact of depth quotes and order sizes", *Contemporary Accounting Research*, Vol. 22 No. 4, pp. 829-865.
- Hermalin, B. and Weisbach, M. (2012), "Information disclosure and corporate governance", The Journal of Finance, Vol. 67 No. 1, pp. 195-233.
- Hong, H., Kubik, J. and and Scheinkman, J. (2012), "Financial constraints on corporate goodness", NBER Working Paper.
- Hou, K. and Moskowitz, T.J. (2005), "Market frictions, price delay, and the cross-section of expected returns", Review of Financial Studies, Vol. 18 No. 3, pp. 981-1020.

- Hu, J., Li, A.Y. and Zhang, F. (2014), "Does accounting conservatism improve the corporate information environment?", Journal of International Accounting, Auditing and Taxation, Vol. 23 No. 1, pp. 32-43.
- Huang, X.B. and Watson, L. (2015), "Corporate social responsibility research in accounting", Journal of Accounting Literature, Vol. 34, pp. 1-16.
- Huang, Y., Liu, L. and Qi, E. (2016), "The dynamic decision in risk-averse complementary product manufacturers with corporate social responsibility", *Kybernetes*, Vol. 45 No. 2, pp. 244-265.
- Hung, M., Shi, J. and Wang, Y. (2013), "The effect of mandatory CSR disclosure on information asymmetry: evidence from a quasi-natural experiment in China", Asian Finance Association (AsFA) 2013 Conference.
- Illia, L., Romenti, S., Rodríguez-Cánovas, B. (2017), "Exploring corporations' dialogue about CSR in the digital era", Journal of Business Ethics, Vol. 146 No. 1, pp. 39-58.
- Jenkins, D.S., Kane, G.D. and Velury, U. (2009), "Earnings conservatism and value relevance across the business cycle", Journal of Business Finance and Accounting, Vol. 36 Nos 9/10, pp. 1041-1058.
- Jian, M. and Lee, K.W. (2015), "CEO compensation and corporate social responsibility", Journal of Multinational Financial Management, Vol. 29, pp. 46-65.
- Jiao, Y. (2011), "Corporate disclosure, market valuation, and firm performance", Financial Management, Vol. 40 No. 3, pp. 647-676.
- Juan, M.G.L., Beatriz, G.O. and Fernando, P. (2014), "Information consequences of accounting conservatism", European Accounting Review, Vol. 23, pp. 173-198.
- Karsalari, A.R., Aghaee, M. and Ghasemi, F. (2017), "The effect of mandatory corporate social responsibility on accounting conservatism in Tehran stock exchange corporations", *International Journal of Academic Research in Accounting, Finance and Management Sciences*, Vol. 7 No. 3, pp. 120-125.
- Khan, M. and Watts, R.L. (2009), "Estimation and empirical properties of a firm-year measure of accounting conservatism", *Journal of Accounting and Economics*, Vol. 48 Nos 2/3, pp. 132-150.
- Kim, J.B. and Zhang, L. (2016), "Accounting conservatism and stock price crash risk: firm-level evidence", Contemporary Accounting Research, Vol. 33 No. 1, pp. 412-441.
- Kim, Y., Li, H. and Li, S. (2014), "Corporate social responsibility and stock price crash risk", Journal of Banking and Finance, Vol. 43, pp. 1-13.
- Kim, Y., Park, M.S. and Wier, B. (2012), "Is earnings quality associated with corporate social responsibility?", The Accounting Review, Vol. 87 No. 3, pp. 761-796.
- Knez-Riedl, J., Mulej, M. and Dyck, R.G. (2006), "Corporate social responsibility from the viewpoint of systems thinking", Kybernetes, Vol. 35 Nos 3/4, pp. 441-460.
- Kurniawan, M. and Wibowo, D.H. (2009), "Analysis on accounting conservatism and CSR disclosures of indonesian banks listed on IDX from 2004 to 2007", *Journal of Applied Finance and Accounting*, Vol. 2, pp. 13-30.
- Labelle, R., Gargouri, R.M. and Francoeur, C. (2010), "Ethics, diversity management, and financial reporting quality", *Journal of Business Ethics*, Vol. 93 No. 2, pp. 335-353.
- LaFond, R. and Watts, R. (2008), "The information role of conservatism", The Accounting Review, Vol. 83 No. 2, pp. 447-478.
- Lambert, R., Leuz, C. and Verrecchia, R.E. (2007), "Accounting information disclosure and the cost of capital", Journal of Accounting Research, Vol. 45 No. 2, pp. 385-420.
- Langberg, N. and Sivaramakrishnan, K. (2008), "Voluntary disclosures and information production by analysts", *Journal of Accounting and Economics*, Vol. 46 No. 1, pp. 78-100.
- Lara, J.M.G., Osma, B.G. and Penalva, F. (2007), "Board of directors' characteristics and conditional accounting conservatism: Spanish evidence", European Accounting Review, Vol. 16 No. 4, pp. 727-755.
- Lara, J.M.G., Osma, B.G. and Penalva, F. (2016), "Accounting conservatism and firm investment efficiency", Journal of Accounting and Economics, Vol. 61 No. 1, pp. 221-238.

Larcker, D.F. and Rusticus, T.O. (2010), "On the use of instrumental variables in accounting research", Journal of Accounting and Economics, Vol. 49 No. 3, pp. 186-205.

- Lee, C., Palmon, D. and Yezegel, A. (2018), "The corporate social responsibility information environment: examining the value of financial analysts' recommendations", *Journal of Business Ethics*, Vol. 150 No. 1, pp. 279-301.
- Leuz, C. and Verrecchia, R.E. (2000), "The economic consequences of increased disclosure", Journal of Accounting Research, Vol. 38, pp. 91-124.
- Leuz, C., Nanda, D. and Wysocki, P.D. (2003), "Earnings management and investor protection: an international comparison", *Journal of Financial Economics*, Vol. 69 No. 3, pp. 505-527.
- Lin, L. (2016), "Institutional ownership composition and accounting conservatism", Review of Quantitative Finance and Accounting, Vol. 46 No. 2, pp. 359-385.
- Lins, K.V., Servaes, H. and Tamayo, A. (2017), "Social capital, trust, and firm performance: the value of corporate social responsibility during the financial crisis", *The Journal of Finance*, Vol. 72 No. 4, pp. 1785-1824.
- McCarthy, S., Oliver, B. and Song, S. (2017), "Corporate social responsibility and CEO confidence", Journal of Banking and Finance, Vol. 75, pp. 280-291.
- McWilliams, A. and Siegel, D. (2000), "Corporate social responsibility and financial performance: correlation or misspecification?", *Strategic Management Journal*, Vol. 21 No. 5, pp. 603-609.
- Michaels, A. and Grüning, M. (2018), "The impact of corporate identity on corporate social responsibility disclosure", *International Journal of Corporate Social Responsibility*, Vol. 3, 3 No. 1.
- Moon, J. and Shen, X. (2010), "CSR in China research: salience, focus and nature", *Journal of Business Ethics*, Vol. 94 No. 4, pp. 613-629.
- Ng, J. (2011), "The effect of information quality on liquidity risk", *Journal of Accounting and Economics*, Vol. 52 Nos 2/3, pp. 126-143.
- Nguyen, P.A., Kecskés, A. and Mansi, S. (2017), "Does corporate social responsibility create shareholder value? The importance of long-term investors", *Journal of Banking and Finance*, Vol. 112.
- Nikolaev, V.V. (2010), "Debt covenants and accounting conservatism", *Journal of Accounting Research*, Vol. 48 No. 1, pp. 137-176.
- Orbik, Z. and Zozulaková, V. (2019), "Corporate social and digital responsibility", *Management Systems in Production Engineering*, Vol. 27 No. 2, pp. 79-83.
- Pan, L.H., Lin, C.T., Lee, S.C. and Ho, K.C. (2015), "Information ratings and capital structure", *Journal of Corporate Finance*, Vol. 31, pp. 17-32.
- Petrenko, O.V., Aime, F. and Ridge, J. (2016), "Corporate social responsibility or CEO narcissism? CSR motivations and organizational performance", Strategic Management Journal, Vol. 37 No. 2, pp. 262-279.
- Prommin, P., Jumreornvong, S. and Jiraporn, P. (2014), "The effect of corporate governance on stock liquidity: the case of Thailand", *International Review of Economics and Finance*, Vol. 32, pp. 132-142.
- Pyo, G. and Lee, H.-Y. (2013), "The association between corporate social responsibility activities and earnings quality: evidence from donations and voluntary issuance of CSR reports", *Journal of Applied Business Research (JABR)*, Vol. 29 No. 3, pp. 945-962.
- Regina, W.M. (2008), "The role of information asymmetry and financial reporting quality in debt trading: evidence from the secondary loan market", *Journal of Accounting and Economics*, Vol. 46, pp. 240-260.
- Riyat, S. (2016), "Enhancement of CSR through digital technology: a futuristic approach", *International Journal of Research in Engineering, Technology and Science*, Vol. 6, p. 7.
- Roulstone, D. (2003), "Analyst following and market liquidity", *Contemporary Accounting Research*, Vol. 20 No. 3, pp. 552-578.
- Salewski, M. and and Zulch, H. (2014), "The association between corporate social responsibility (CSR) and earnings quality: evidence from European blue chips", HHL Working Papers.

Accounting conservatism

- Sheehy, B. (2015), "Defining CSR: problems and solutions", Journal of Business Ethics, Vol. 131 No. 3, pp. 625-648.
- Sheu, H.J., Chung, H. and Liu, C.L. (2010), "Comprehensive disclosure of compensation and firm value: the case of policy reforms in an emerging market", *Journal of Business Finance and Accounting*, Vol. 37 Nos 9/10, pp. 1115-1144.
- Shroff, N., Sun, A.X., White, H.D. and Zhang, W. (2013), "Voluntary disclosure and information asymmetry: evidence from the 2005 securities offering reform", *Journal of Accounting Research*, Vol. 51 No. 5, pp. 1299-1345.
- Tang, Y., Mack, D.Z. and Chen, G. (2018), "The differential effects of CEO narcissism and hubris on corporate social responsibility", Strategic Management Journal, Vol. 39 No. 5, pp. 1370-1387.
- Tsao, S.M., Lu, H.T. and Keung, E.C. (2016), "Voluntary monthly earnings disclosures and analyst behavior", *Journal of Banking and Finance*, Vol. 71, pp. 37-49.
- Vanstraelen, A., Zarzeski, T. and Robb, W.G. (2003), "Corporate nonfinancial disclosure practices and financial analyst forecast ability across three european countries", *Journal of International Financial Management and Accounting*, Vol. 14 No. 3, pp. 249-280.
- Wang, L.F.S., Sun, J. and Liu, Y.S. (2019), "Institutional pressure, ultimate ownership, and corporate carbon reduction engagement: evidence from China", *Journal of Business Research*, Vol. 104, pp. 14-26.
- Watts, R. (2003), "Conservatism in accounting part I: explanations and implications", Accounting Horizons, Vol. 17 No. 3, pp. 207-221.
- Welker, M. (1995), "Disclosure policy, information asymmetry, and liquidity in equity markets", Contemporary Accounting Research, Vol. 11 No. 2, pp. 801-827.
- Yang, S., He, F., Zhu, Q. and Li, S. (2018), "How does corporate social responsibility change capital structure?", Asia-Pacific Journal of Accounting and Economics, Vol. 25 Nos 3/4, pp. 352-387.
- Ye, K. and Zhang, R. (2011), "Do lenders value corporate social responsibility? Evidence from China", *Journal of Business Ethics*, Vol. 104 No. 2, pp. 197-206.
- Young, K.K. (2005), "Accounting conservatism and managerial incentives", Management Science, Vol. 51, pp. 1626-1632.

Corresponding author

Kung-Cheng Ho can be contacted at: z0004531@zuel.edu.cn